

Upjohn Press

Upjohn Research home page

11-28-2022

Long-Term Care in the United States: History, Financing, and **Directions for Reform**

George A. (Sandy) Mackenzie

Follow this and additional works at: https://research.upjohn.org/up_press



Part of the Labor Economics Commons

Citation

Mackenzie, George A. (Sandy). 2022. Long-Term Care in the United States: History, Financing, and Directions for Reform. Kalamazoo, MI: W.E. Upjohn Institute for Employment Research. https://doi.org/ 10.17848/9780880996891



This work is licensed under a Creative Commons Attribution-NonCommercial-Share Alike 4.0 International License.

This title is brought to you by the Upjohn Institute. For more information, please contact repository@upjohn.org.

Long-Term Care in the United States

HISTORY, FINANCING, AND DIRECTIONS FOR REFORM

George A. (Sandy) Mackenzie



Long-Term Care in the United States

Long-Term Care in the United States

History, Financing, and Directions for Reform

George A. (Sandy) Mackenzie

2022



W.E. Upjohn Institute for Employment Research Kalamazoo, Michigan

Library of Congress Cataloging-in-Publication Data

Names: Mackenzie, G. A. (George A.), 1950- author.

Title: Long-term care in the United States: history, financing, and directions for reform / George A. (Sandy) Mackenzie.

Description: Kalamazoo, Michigan: W.E. Upjohn Institute for Employment Research, 2022. | Series: WE focus series | Includes bibliographical references and index. | Summary: "This book is a concise survey of the development of U.S. long-term care and its financing, with comparisons with other rich countries. It also includes a brief comparative account of the impact of the COVID-19 pandemic in the United States and several other countries. The study finds much that is amiss with American long-term care and proposes three sets of progressively more ambitious reforms"— Provided by publisher.

Identifiers: LCCN 2022045038 (print) | LCCN 2022045039 (ebook) | ISBN 9780880996884 (paperback) | ISBN 9780880996891 (ebook)

Subjects: LCSH: Long-term care facilities—United States. | Nursing home care—United States. | Long-term care of the sick—United States.

Classification: LCC RA997 .M33 2022 (print) | LCC RA997 (ebook) | DDC 362.160973—dc23/eng/20221108

LC record available at https://lccn.loc.gov/2022045038

LC ebook record available at https://lccn.loc.gov/2022045039

© 2022

W.E. Upjohn Institute for Employment Research 300 S. Westnedge Avenue Kalamazoo, Michigan 49007-4686

The facts presented in this study and the observations and viewpoints expressed are the sole responsibility of the author. They do not necessarily represent positions of the W.E. Upjohn Institute for Employment Research.

Cover design by Carol A.S. Derks. Index prepared by Diane Worden. Printed in the United States of America. Printed on recycled paper.

Contents

A	cknowledgments	vii
In	troduction	1
1	A Brief History of Long-Term Care	5
	The Establishment of Nursing Homes	7
	The Current Regulatory Framework for Nursing Homes	8
	The Advent of Home- and Community-Based Services	11
	Assisted Living Facilities	14
	Continuing Care Retirement Communities	16
	The Long-Term Care Paid Labor Force	18
	The Department of Veterans Affairs' Long-Term Care Program	22
	Recent Developments in the Provision of Long-Term Care at	23
	the State Level	
2	How Care Is Financed	31
	Medicaid	31
	Eligibility for Medicaid's Institutional and Home- and Community-Based Services Coverage	33
	Actuarial and Financial Aspects of Continuing Care Retirement Communities	39
	Private Long-Term Care Insurance	42
	Unpaid Caregivers	45
3	A Comparative Analysis	53
	Some Consequences of the Means-Tested Approach	54
	How Canada and Germany Address the Challenge of Long- Term Care	57
	Lessons for the United States	62
4	The Impact of the COVID-19 Pandemic on Older Americans	63
	in Nursing Homes and Other Institutions	
	Nursing Homes	65
	Experiences in Other Countries	68
	Assisted Living Facilities and Continuing Care Retirement Communities	70

5	A Thumbnail Sketch of a Good System How Does the Current System Compare to the Benchmark? Directions for Reform	73 75 77
Ref	Terences	85
Aut	chor	91
Ind	ex	93
Abo	out the Institute	103
	Tables	
1.1	Residents in Long-Term Care Institutions and Persons Cared for at Home	6
1.2		19
1.3	,	19
1.4	Indicators of Poverty in the Direct Care Workforce, 2020 (% of group population)	20
2.1	Number of States with the Special Income Pathway and/or the Medically Needy Pathway	37
3.1	Long-Term Care Expenditure as a Percentage of GDP, G-7 Countries	56
4.1	Pandemic-Related Nursing Home Deaths in the United States, the United Kingdom, and Canada (March 2020–April 2021)	69

Acknowledgments

Many friends and colleagues, and even two relatives, have helped me with this project by suggesting important studies on long-term care and by commenting on earlier drafts: Bruce Cain, Dr. Margaret Casey, Jack Cumming, John Cutler, Annie Deal, Dan Doonan, Jeffrey Fine, Howard Gleckman, Dan Hewitt, Julia Hewitt, Charles Jeszeck, George Macintosh, Margie Mackenzie, Linda Mallon, Steve Newman, Kerry Pechter, Anna Rapapport, Susan Reinhard, Gerd Schwartz, Meyer Statman, Eileen Tell, Jack Towarnicky, Joe Tomlinson, Ben Veghte, Tony Verheyen, and the Upjohn Institute's external reviewer. Despite all the help and support these ladies and gentlemen have given me, the usual disclaimer must apply in full force. Any errors, be they of logical inference, of fact, or errors of any other kind that remain in this study are my responsibility and mine alone.

In memory of my mother-in-law, Ruth Medernach.

And in memory of Jonathan Barry Forman—gentleman, scholar, friend.

"That is no country for old men."

-W.B. Yeats, Sailing to Byzantium

"Sir, you are old. Nature in you stands on the very verge of its confines . . . being weak, seem so."

-William Shakespeare, King Lear, Act II, Scene 4

" \dots the test of a civilization is the way that it cares for its helpless members."

—Pearl Buck

Introduction

If we judge our society by the care it provides its most vulnerable members, we must find it wanting in some respects. This study finds that much is amiss with the provision and financing of long-term care for elderly people in the United States with respect to both institutional care in nursing homes and care provided in the home.

In the United States, long-term care is provided in congregate facilities and in homes. Home care is by provided by individuals sponsored under home- and community-based service (HCBS) programs financed by Medicaid, a government program financed by the federal government and the states, and by caregivers, many of whom are relatives of the persons being cared for and most of whom are unpaid. The number of elderly persons being cared for at home was estimated in 2019–2020 to be 52.0 million, dwarfing the 2.9 million at congregate facilities.

There are three types of congregate facilities in the United States: nursing homes, which typically offer only semiprivate accommodation (two to a room); assisted living facilities (ALFs), which provide separate apartments for their residents; and continuing care retirement communities (CCRCs), which provide apartments and homes for their residents. ALFs and CCRCs cater to an affluent clientele. Nursing home care is financed largely by Medicaid; however, Medicaid does not finance room and board at ALFs, and plays no role in the financing of CCRCs.

This study identifies six major weaknesses with the current U.S. system:

- Long-term care coverage is severely limited by the stringent asset and income tests that Medicaid requires for eligibility for nursing homes and HCBS. As a result, the United States does not compare well with other countries in its coverage of the elderly infirm or of their caregivers.
- 2) Despite their outsized role, caregivers of the elderly at home receive little governmental support.

- 3) Basic information on some institutions or parts of the longterm care system is not always timely or comprehensive.
- 4) Medicaid is complex and hard to understand. Many Americans think long-term care is financed by Medicare, which covers only short-term stays. This lack of understanding could leave them unprepared to deal with Medicaid's stringent eligibility tests if and when they confront the issue of financing long-term care.
- Incentives to rectify shortcomings in the care provided by congregate facilities, especially nursing homes, may be inadequate.
- 6) The low wages for paid caregivers are a matter of concern.

The chapters that follow in this book assess long-term care in all its aspects, not just institutional care. Chapter 1 discusses how the emergence of nursing homes in the 1930s replaced the poorhouses that had previously housed the indigent elderly. It then addresses the introduction of Medicaid in 1965 and its initial bias toward nursing homes as opposed to care at home. It provides a discussion about the advent of HCBS and the growth of ALFs and CCRCs. Chapter 1 also includes a section on paid caregivers.

Chapter 2 considers the financial arrangements governing the provision of long-term care. In particular, the section on Medicaid explains how the federal government interacts with the states to finance both institutionally based care and HCBS, and the criteria that determine eligibility for Medicaid and share of the costs that it covers. The chapter also discusses the anemic market for private long-term care insurance.

Chapter 3 provides a comparative analysis of long-term care in member countries of the Organisation for Economic Co-operation and Development (OECD). It briefly describes the various models of provision of long-term care and provides thumbnail sketches of the way long-term care is delivered and financed in Canada and in Germany, a country that has been cited as a possible model for the United

States. The Canadian system also has some features that, if adopted, might increase the coverage and also the public financing of longterm care in the United States. The comparative analysis of Chapter 3 is not flattering to the U.S. system.

Chapter 4 addresses the impact of the pandemic on nursing homes in the United States and includes some comparison with Canada, Germany, and the United Kingdom, three other countries for which data are readily available. The losses in U.S. nursing homes before the roll-out of vaccination programs in the winter and spring of 2021 were staggering, and were higher than those of the comparator countries, particularly Canada and Germany.

Finally, Chapter 5 sets out a benchmark for a good system. It compares the U.S. system to it and, as this chapter has already highlighted, finds that the current system falls short of the proposed benchmark in important respects. Chapter 5 concludes by offering three possible directions for reform that are progressively more ambitious but also more politically contentious.

Chapter 1

A Brief History of Long-Term Care

The modern history of long-term care in the United States begins with the establishment of nursing homes as distinct institutions in the 1930s. The next major development was the advent of public support for home- and community-based services (HCBS) in the 1970s. Recent decades have seen the rise in popularity of assisted living facilities (ALFs), which, unlike nursing homes, are typically private pay. Continuing care retirement communities (CCRCs), which, like ALFs, cater to relatively affluent elder Americans began to flourish in the 1970s, and their numbers have grown substantially since then.

The regulatory role of states and the federal government is greatest for nursing homes and HCBS, but it is more limited for ALFs and more limited still for CCRCs. Reflecting this difference in the degree of regulatory oversight, the information available on nursing home residents and the quality of the care they receive is far greater than that on the residents in ALFs and CCRCs. The most recent development in long-term care has been moves by several states toward implementing their own programs of long-term care; Washington State, for example, has introduced a system that is complementary to Medicaid.

Studies of long-term care typically focus on the care that is paid for at least partially by Medicaid, whether provided at nursing homes or through the community. However, these studies are limited by the fact that the number of elderly and disabled Americans who are cared for at home by their families dwarfs the number of elderly Americans living in one of the three types of congregate institutions (see Table 1.1).

Although some of the elderly living at home with family benefit from Medicaid-financed services, caregivers who are relatives

ior	at Home			
		Assisted	Continuing	
		living	care retirement	į.
	Nursing	facilities	communities	Persons cared
	homes	(ALFs)	(CCRCs)	for at home
Year reported	2019	2019	2020	2020
Residents	1,246,000	812,000	800,000	52,000,000

Table 1.1 Residents in Long-Term Care Institutions and Persons Cared for at Home

NOTE: Some of these statistics are quite uncertain. ALFs do not routinely report resident numbers to their state. The CCRC figure is an educated guess, and the figure for persons being cared for at home is based on a survey.

SOURCE: Nursing homes: Kaiser Family Foundation (2020); ALFs: National Center for Health Statistics (2019) Table VIII; CCRCs: Various sources; Persons cared for at home: AARP (2020a) and author's estimate.

and most other caregivers are almost all unpaid. The total number of caregivers of adults aged 50 years or older in 2019 has been estimated to be as large as 42 million, and the number of adults they care for exceeds that number—an estimated 52 million, a number that is projected to be about 20 times the number in nursing homes, ALFs, and CCRCs combined.² Typically, the care recipient is a parent of the caregiver with an average age of 75. Some 45 percent of care recipients are estimated to have multiple health conditions or disabilities (AARP 2020a).³

The number of elderly Americans cared for at home is certain to have grown in 2020–2021 in response to the fear of contracting COVID-19 in congregate or institutional settings, especially nursing homes, where occupancy fell by 16 percent between January 2020 and January 2021 (Cottle 2021a). The aging of the baby boom generation will continue to have the same effect. This chapter does not directly address the history of stay-at-home care, but it is important to be mindful of the huge role it plays in long-term care. The next chapter will address the broader economic and financial implications of unpaid caregiving.

THE ESTABLISHMENT OF NURSING HOMES

Before the Social Security Act (SSA) of 1935, most disabled or ill elderly who were not cared for at home were lodged in countyor city-run poorhouses, where conditions were often frightful. A few may have had the comparatively good fortune of living in homes that were established for their religious sect or ethnic community, although even here conditions might be far from luxurious. As Smith and Feng (2010, p. 2) put it, "Long-term care was, in essence, the last holdover of the Elizabethan poor-law approach."4

The Old Age Assistance program of the SSA led to the creation of the private nursing home industry by providing funds to states that established residences specifically for the elderly poor while prohibiting funding for residents in poorhouses, a policy that emptied out the poorhouses, at least of their elderly residents. In 1950, the funds from Old Age Assistance began to be paid directly to the nursing homes themselves, rather than to their residents, and nursing homes had to be certified by their state to be eligible for funding.⁵ In the mid-1950s, the Hill-Burton Act provided some funding for the construction of nursing homes.6

In 1965, Medicare and Medicaid were passed as amendments to the SSA. Medicare, in addition to providing health insurance for Americans aged 65+, covers nursing home stays for short-term or acute care. Medicaid covers long-term stays and unlike Medicare is jointly funded by the federal government and the states. Initially, Medicaid covered only the indigent, blind, and disabled. In addition to being poor, those elderly who were not blind or disabled had to demonstrate a need for assistance with at least one of the activities of daily living (ADLs).7 In this respect, it might be said that Medicaid was perpetuating the poorhouse approach to the care of the elderly.

Medicaid's provisions are extremely complex. They vary not only from one state to another, but within states depending on the programs offered. In addition to institutional services, the states are

obliged to offer certain home health services to qualifying state residents—namely, part-time or occasional nursing services; home health aide services; and medical supplies, equipment, and appliances suitable for use in the home to qualifying Americans aged 65 or older (Watts et al. 2020).8

Other services may be provided at the option of the individual state, and the same state may have different programs covering essentially the same services. As will be explained in greater detail in Chapter 2, Medicaid's coverage was subsequently expanded by the addition of two pathways to the original pathway established by the Johnson administration. The complexity of Medicaid's provisions may partly explain why many Americans believe that long-term care is covered by Medicare, when actually that program covers only short-term postsurgical stays and the like.

Medicaid's initial bias in favor of institutions and against community-based care has been interpreted as an effort to avoid policies that would entail moral hazard, since a genuine need for home care is harder to monitor and ascertain than the need for institutionalized care (Smith and Feng 2010).⁹

THE CURRENT REGULATORY FRAMEWORK FOR NURSING HOMES

The subsequent history of the Medicaid program has been marked by a series of legislative acts to toughen the oversight and improve the standard of care in nursing homes. These acts at times have been enacted in response to publicly voiced concern over the quality of care in the homes. In 1968, the Department of Health, Education and Welfare (subsequently renamed the Department of Health and Human Services), which had assumed overall responsibility for Medicaid's administration, was authorized to set standardized regulations for care and withhold funding from homes that failed to achieve the stipulated standards of care. ¹⁰ The Nursing Home Reform Act of 1987 strength-

ened federal standards, inspections, and enforcement provisions; set uniform standards for nursing homes funded by either Medicare and Medicaid; required comprehensive resident assessments; set minimal requirements for licensed nursing staff; and required inspections to focus on care outcomes.

The Affordable Care Act of 2010 included requirements for facility quality assurance, performance improvement programs, and compliance and ethics programs. In 2016, the Obama administration acted to implement these requirements and introduced revamped standards for the assessment of the quality of care of nursing home residents, including those related to infection control, facility assessment, and emergency preparedness. Among other reforms, the 2016 regulations also revised provisions to give more attention to person-centered care (i.e., care that is more tailored to individual residents) and the reporting of abuse and neglect. Additionally, the 2016 regulations added a new section on behavioral health services and required staff competency vetting to determine staffing sufficiency as well as new staff training requirements. The regulations were implemented in three phases between 2016 and 2019 (Musumeci and Chidambaram 2020).

The current regulatory framework requires states to oblige certified nursing homes to provide them with a very substantial amount of information annually on the condition of nursing home residents and the quality of care they receive, in addition to data on the number of beds, residents, and ownership status and size of homes. Nursing homes must also provide data on the number of their residents unable to perform any one or more of the six ADLs, as well as data on basic indicators of health and well-being, such as the incidence of incontinence, pain, and pressure ulcers (bed sores). This information is passed on to the federal government and is published by various agencies.11

Improving both the quality of information regarding various dimensions of the quality of care in nursing homes and the effectiveness of measures to rectify problems once they are found has been an ongoing challenge since the beginning of federal and state oversight.

Over the years, several legislative initiatives whose goals included these basic objectives have failed to pass through Congress.¹²

A Summary of Annually Reported Informational Requirements for Nursing Homes with Some Basic Data

Ownership categories, funding of care, and resident population: In 2015–2016, of the nation's 15,600 nursing homes, 69 percent were for-profit, 23 percent were nonprofit private, and 7 percent were publicly owned (National Center for Health Statistics [NCHS] 2019). These shares have not varied enormously in the past 20 years. The nation's 1.4 million nursing home residents as of the 2016 count had declined to 1.2 million as of 2019, according to the Kaiser Family Foundation (2020). Of the total number of residents, 62 percent were funded by Medicaid, 14 percent by Medicare (for short-term stays), and 25 percent by private or other sources (Harrington et al. 2018). As Chapter 4 notes, the number of residents has fallen substantially since 2016 both because of the deaths caused by the COVID-19 pandemic and decisions by potential residents and their families to opt for home care or assisted living.

Quality of care: Nursing homes are required to report shortcomings over a comprehensive list of indicators of the quality of care, considering both the frequency and the severity of the problem. A finding of a specific problem is known as a health deficiency. Over the 2005–2014 period, the most cited health deficiencies pertained to food sanitation, accident prevention, overall quality of care, and infection control (CMS 2015). Depending on the severity and prevalence of deficiencies, a nursing home may lose its state certification. The imposition of such a draconian penalty is rare. In addition to health deficiencies, nursing homes are required to report instances of abuse by nursing home aides, be it physical, mental, or emotional.

A recent report from the Government Accountability Office (GAO 2020) that analyzed data from the Centers for Medicare & Medic-

aid Services (CMS) for 2013–2017 found that 82 percent of nursing homes had been cited for an infection control deficiency. About half of these were cited for one or more consecutive years during this period. However, only 1 percent of the deficiencies were found to be severe, in the sense of posing immediate harm to one or more residents.¹⁴ Another GAO report (2019) on instances of abuse in the same period found that although the incidence was low at the beginning of the period under study, it rose dramatically over the five years. The report also found that information on the type of abuse or its perpetrator was not readily available; therefore, it is fair to conclude that although reporting of health deficiencies is adequate, follow-up appears to be lacking, and the situation for abuse is likely much worse.

THE ADVENT OF HOME- AND COMMUNITY-BASED SERVICES

In the 1970s, efforts began to authorize the use of federal funds for noninstitutional HCBS. In 1970, Medicaid funds could be used for the first time to finance noninstitutional care for the elderly with functional disabilities if those deficiencies would qualify them for nursing home residence (Iezzoni, Gallopyn, and Scales 2019). The SSA was amended to permit federal grants to states for social services programs such as homemaker services, adult day care, and health support.¹⁵ In 1975, amendments to the SSA created Title XX, which consolidated federal assistance to states for social services into a single grant, giving states more flexibility in the allocation of the funds across different programs under the general rubric of HCBS while requiring states to prevent or reduce inappropriate institutional care by providing for these at-home services.

Potential problems with moral hazard aside, home care should in principle impose less of a burden on public finances than institutionalized care, because the beneficiaries of home care remain in their homes, sparing the federal government and the states the cost of lodging and feeding them. One study has found that states with established HCBS programs achieved lower costs than other states (Kaye, LaPlante, and Harrington 2009). Medicaid expenditure on HCBS began to grow significantly in the 1980s, but waiting lists for the optional HCBS that states offer have lengthened because states can limit the number of persons eligible for the programs offered.¹⁶

The Americans with Disabilities Act, enacted in 1990, emphasized the importance of integrating people with disabilities into the community and ending exclusion and segregation. Another step was the creation of the Money Follows the Person demonstration program, which aimed to support state efforts to rebalance their long-term services and supports system to emphasize HCBS by giving individuals a choice of where they live and receive services. From the program's authorization in 2005 until 2021, states transitioned more than 100,000 people to community living, amounting to about 7 percent of the number of nursing home residents as of 2008. About one-third of those transitioned were elderly persons; the rest were disabled younger persons. In addition to its obvious benefits for the individuals who returned to the community, the program has saved Medicaid money (Gottlich 2021).

The Affordable Care Act provided additional choices to states aimed at encouraging the improvement of their long-term care infrastructures and expanding HCBS. Its provisions included an extension of the Money Follows the Person program. In addition, for the five-year period beginning January 1, 2014, states were required to apply spousal impoverishment standards—which had been introduced earlier to avoid the impoverishment of the spouses of persons needing institutionalized care—in determining eligibility for married Medicaid applicants receiving HCBS. Prior to this, these standards were applied only to the spouses of nursing home residents.¹⁷

The number of elderly Americans benefiting from adult day service centers, which is one component of HCBS, was about 300,000 in 2016 (NCHS 2019), with Medicaid financing about two-thirds of

the total cost. By way of comparison, home health agencies—which numbered 12,200 in 2015-2016, and which are almost entirely financed by private health and/or long-term care insurance—had about 4.5 million users in 2016 as measured by the number of patients discharged in that year (NCHS 2019). A substantial share of expenditure on HCBS goes to paying for home visits by doctors and other medical personnel, as well as visits by aides who assist with ADLs. As of fiscal year 2018 (fiscal years end on September 30 of the year stated), the states provided HCBS to about 620,000 elderly persons under their basic Medicaid obligation. An additional 4.2 million benefited from optional programs (Watts, Musumeci, and Chidambaram 2020).18 With the increased emphasis on HCBS, the share of Medicaid's budget going to institutional care has declined; spending on HCBS surpassed spending on institutional care for the first time in 2013. In 2016, it comprised 57 percent of total Medicaid Long-Term Services and Supports spending (Watts, Musumeci, and Chidambaram 2020).

In addition to paying for in-home visits by doctors and other providers of medical services, a state's Medicaid program may also pay for the services of certain family and non-family members who assist Medicaid-eligible persons with ADLs. In most states, spouses of eligible persons may not be employed in this capacity, but divorced spouses may. Rates of remuneration vary from state to state but are generally at the minimum wage level. Data on the relative roles of paid and unpaid caregivers are not readily available, although it appears that the great majority of these caregivers' time is unpaid (AARP 2020a).19

A discussion of HCBS should not conclude without a reference to the Programs for All-Inclusive Care for the Elderly (PACE). PACE began as a single center in San Francisco's Chinatown in the early 1970s aimed at providing home services to older adults who qualified for institutional care but with adequate care could avoid being institutionalized. The original day care center began receiving Medicaid payments in 1974. Under the program, each participant benefits from

care by a multidisciplinary team. The PACE model gradually spread across the country, with each organization acting as a separate non-profit covering a particular geographical area at a single physical site. In 1990, the first PACE programs received Medicare and Medicaid waivers to operate. By 2019, 130 PACE organizations were operational in 31 states and were serving over 50,000 participants (National Pace Association, n.d.). A fixed sum is paid for each participant, which allows participants to benefit from services not covered under Medicaid's fee-for-service approach. PACE is an optional program under Medicaid: that is, states are not required to offer it.

ASSISTED LIVING FACILITIES

The total number of ALFs in the United States as of 2019 is estimated to be nearly 29,000, with a capacity of about one million beds and number of residents about 812,000 (NCHS 2019). The average number of residents per facility was 28, compared with 86 at the nation's nursing homes. Some 57 percent of the nation's ALFs are chain-affiliated (i.e., related to a group of facilities under the same ownership) (NCHS 2019).

Assisted living centers cater to more affluent elderly people than nursing homes do and are intended for those elderly who want the privacy of their own apartment but typically cannot perform all the ADLs. Apartment units normally have a small kitchen area with a fridge and a microwave for snacks and reheated meals, but meals are usually served in a communal dining room. The residents in ALFs do tend to need less assistance with the ADLs than nursing home residents, and many residents in ALFs may be quite mobile. For example, in 2015–2016, only one in five residents in ALFs and similar residential care communities needed assistance with eating, compared with 6 in 10 nursing home residents. Ninety percent or more of nursing home residents needed assistance with one or more of the other ADLs, compared with 40–60 percent of residents in ALFs and

other care communities. However, ALFs usually also include what is known as a memory unit for residents suffering from Alzheimer's or other dementias, and some 42 percent of residents were diagnosed with one of these conditions, not far below the 48 percent of nursing home residents with that diagnosis (NCHS 2019).

Although residents in ALFs have more living space than the typical nursing home resident, ALFs are less expensive than nursing homes. In 2020, the nationwide median annual cost of a private one-bedroom unit in an ALF was estimated to be \$48,000, compared to about \$93,000 for a semiprivate unit in a nursing home (American Health Care Association/National Center for Assisted Living 2021). The ALF estimate, however, does not include charges for such extra services as the supply and monitoring of medications, which can exceed \$1,000 per month.

A comparison of the make-up of the labor force at ALFs and nursing homes may help explain part of the difference in the cost of the two types of institutions. At nursing homes, nurses at all levels of training accounted in 2016 for 34 percent of the full-time equivalent labor force; at ALFs they make up 16 percent (NCHS 2019, Table VI). Aides, mostly unskilled and low-paid, account for a correspondingly higher percentage of the full-time equivalent labor force at ALFs. The average hourly wage earned by certified nursing assistants is about \$14 (Bureau of Labor Statistics 2021). Cottle (2021b) implies that this rate is on a par with the wage paid by Chipotle, the fast-food chain, and it is certainly less than what a nurse would receive.

Similar to the regulation of nursing homes, the regulation of ALFs is split between the federal and state governments. The federal government's involvement is exercised through Medicaid. Although Medicaid does not cover room and board, it may cover personal care services for residents who would be eligible for Medicaid were they eligible for Medicaid coverage in a nursing home.

Ideally, Medicaid should ensure that the elderly and disabled receive care in the setting most appropriate for their needs. An estimated 48 percent of ALFs are Medicaid-certified to be home and community-based service providers, and about 17 percent of ALF residents rely on Medicaid to cover some aspect of the cost of their daily care. Although most states offer Medicaid coverage of assisted living services, each state administers its Medicaid programs and has some latitude in the programs it will cover, so that beneficiary eligibility criteria and provider participation range from very limited to robust. In any case, Medicaid does not cover room and board, so it plays a relatively minor role in paying for part of the total cost of residence at an ALF, with less than one in five residents at ALFs being covered even in part by Medicaid in 2016. As a result, ALFs will be beyond the means of most elderly Americans.

The financial arrangements of ALF residents are comparatively straightforward. A new resident enters into a rental agreement with the facility, with a term that is typically month to month. However, residents typically have no legal protection from a decision by the facility not to renew an expiring lease, and there are no limits on rate increases. In practice, it is likely that the lease of a resident will be renewed unless the ALF decides that it is no longer capable of caring for that resident because of a deterioration in their medical condition.

As we shall see in Chapter 4, the greater privacy that residents at ALFs enjoy has held down substantially the total deaths caused by the COVID-19 pandemic in these facilities compared to nursing homes. However, a full picture has yet to emerge because, as of mid-2020, the latest date for which data are available, less than half of the states were reporting COVID-19-related deaths in ALFs.

CONTINUING CARE RETIREMENT COMMUNITIES

In the early 1980s, when the first thorough analysis of CCRCs was published (Winklevoss and Powell 1984), there were about 275 such facilities in the country, with about 70,000 residents. Industry sources have reported that the precursors to these communities date back to the early twentieth century, when religious and other

groups established them for their members, who typically financed their membership by selling their homes (GAO 2010). The number of CCRCs has now grown to nearly 2,000, and the number of their residents is believed to number about 600,000 (AARP 2019). As their name suggests, CCRCs provide residents with services that range from separate apartments or stand-alone dwellings with community amenities, often quite luxurious, for those residents who live independently, to assisted living facilities for members needing help with one or more of the ADLs, to the services of a nursing home. The basic appeal of a CCRC is precisely that residents can in principle be assured of exactly the level of care they need without having to move elsewhere. Consequently, stays at CCRCs may be quite long in comparison with ALFs, particularly if the resident of an ALF is deemed to require care that the facility is incapable of providing.

One issue that crops up with any attempt to study or report on CCRCs is the paucity of basic data on their operations. Even the figure just given for the total number of their residents is an educated guess. The lack of data is most likely due to the fact that there is no federal regulatory framework, and that only 38 states have imposed a regulatory framework of any kind on them. The absence of data may be of less importance than it would be for nursing homes because residents at CCRCs are more affluent than the general population in their age group, are in better health, and are likely better able to advocate for themselves. Nonetheless, CCRCs involve some potentially serious financial risks for their residents.

Most CCRCs have three basic payment arrangements, plus a pay-as-you-go rental arrangement. These arrangements each involve a large entrance fee plus an ongoing monthly fee. The difference among them depends on the treatment of health-related expenditure. The largest up-front fee applies when the monthly fee is not affected by the level of health care the resident needs. Once a person becomes a member of a CCRC requiring an up-front payment, she cannot be asked to leave if her health deteriorates, although depending on the contract she signed upon entering the community, her monthly fee may rise.

In 2019, the entry fee for the CCRCs that have one, which represent close to two-thirds of the total, was an estimated \$329,000, plus a monthly maintenance or service fee of \$2,000–\$4,000. The monthly rent for the remaining CCRCs was estimated at about \$3,000–\$6,000 (AARP 2019).

The financial arrangements of CCRCs are complex, essentially because by accepting someone as a member, the CCRC is committing to ensure care at an appropriate standard for the rest of the member's life. It is thus taking on a complex contingent liability, because neither a resident's length of life nor the quality of care he may need is predictable. The implications of this commitment and the regulatory challenges it poses are explored further in Chapter 2.

THE LONG-TERM CARE PAID LABOR FORCE

The provision of long-term care is a labor-intensive business. According to the Bureau of Labor Statistics (BLS) Division of Occupational Employment Statistics, in May 2020 some 4.6 million persons were employed as direct care workers in nursing homes, other congregate settings, and in the provision of care in community centers and homes. Direct care workers, as the language suggests, provide basic care to elderly and disabled persons in the performance of the basic ADLs as well as other duties. They are normally classified in one of three occupational groups: home health aides, personal care aides, and nursing assistants.²⁰ There is, however, considerable overlap in their duties. They work in one of three settings: private homes; congregate settings, including ALFs and group houses; and nursing homes (see Table 1.2).

Nursing assistants, most of whom work in nursing homes, assist residents with ADLs but may also perform basic clinical duties (like monitoring blood pressure) under the supervision of a nurse. Home health aides and personal care aides work in the home, community centers, and the other congregate settings noted. The sector also

Home care workers	2,400	
Residential care aides	675	
Nursing assistants in nursing homes	527	
Subtotal	3,602	
Direct care workers in other industries	998	
Total	4,600	

SOURCE: PHI (2021).

employs professionally certified staff, of which the largest single occupational group would be registered and practical nurses, as well as doctors and nonmedical professional staff, such as accountants and business analysts.

The three groups are quite similar in their demography: the vast majority are women and persons of color (see Table 1.3). In addition, about 15 percent are non-U.S. citizens (PHI 2021). Hourly wages are low, and median annual incomes are very low compared with average median incomes for the nation's workforce. Part of the reason for this is that many direct care workers work less than a 35-hour week, which makes them part time. Surveys imply that the typical

Table 1.3 Basic Characteristics of the Direct Care Workforce, 2020

	Home care workers	Residential care aides	Nursing home assistants
Number (000s)	2,400	675	527
Demographic (% of total workforc	e)		
Women	86	81	91
Persons of color	63	53	58
With at least one minor child	23	26	31
Median age	47	37	38
Economic (% of total workforce)			
Working part time	42	23	24
Share without health insurance	17	17	13
Median hourly wage (\$)	12.98	13.45	14.48
Median annual income (\$)	18,100	22,200	24,200

SOURCE: PHI (2021).

reason for this is "noneconomic," such as poor health or the need to look after children. Home care workers stand out from the other two groups in two respects: their median age is distinctly higher, as is the share who work part time (see Table 1.3). The higher share of home care workers on part-time schedules helps explain the gap between their annual income and that of the other two groups. The educational attainment of the three groups is similar: about half have no more than high school education.

It is important to note that the estimate of the number of home care workers shown in Tables 1.2 and 1.3 does not include home care workers who are directly employed by the "consumer"—the recipient of the care under Medicaid's consumer-directed programs. PHI, an advocacy group for direct care workers and the source for the data shown in Tables 1.2 and 1.3, estimates that these workers may number 1.2 million, but recognizes that the indirect source of its estimate makes it uncertain (PHI 2021).

In light of the low average incomes of direct care workers, it is not surprising that all three classes tend to reply on public assistance in one form or another (see Table 1.4). That reliance does vary somewhat across the three groups, but for all three it is significant. Equally striking is the share of direct care workers who are members of a household that lives below two times the poverty line. Moreover,

Table 1.4 Indicators of Poverty in the Direct Care Workforce, 2020 (% of group population)

	Home care workers	Residential care aides	Nursing home assistants
Living in a household with income:			
1) Below the poverty line	16	13	12
2) Below 2x the poverty line	45	40	41
Receiving some form of public assistance ^a	53	38	34

^a Medicaid, food, and nutrition (e.g., Supplemental Nutrition Assistance Program or cash).

SOURCE: PHI (2021).

more than 1 in 10 households with a direct care worker live below the poverty line (Table 1.4).

A recent issue brief from the Kaiser Family Foundation (Musumeci, Amula, and Rudowitz 2021) goes beyond the numbers to address the difficulties encountered by both the direct workforce and unpaid family workers. The brief reports on the results of a focus group exercise organized by the Kaiser Family Foundation in the summer of 2021. By its nature, a focus group exercise cannot be deemed to be representative of whole populations. Nonetheless, the sentiments and views reported in this brief are similar to those reported by other sources, including PHI (2020).

Unsurprisingly, direct care workers believe that they are underpaid for the physical and mental demands of their work, although at the same time they take pride in the work they do. Shift workers complain of unpredictable changes in their hours, which were aggravated by the pandemic. Some members of the groups reported that, while they received training at the employer's expense, they did not benefit from wage increases as a result, nor was there any kind of career ladder.21

Medicaid payments to nursing homes and to those home care workers who are paid by Medicaid's HCBS programs appear to put a ceiling on the wages that their direct care workers receive. In spite of reported labor shortages, there has been no increase in real wages, which have in fact declined.²² The demand for home care workers is expected to grow more rapidly than the demand for personal care and nursing aides, in part because of the momentum of the aging in place movement, which has been given a boost by the impact of the pandemic on the perceived attractiveness of nursing homes.²³ Whether this will lead to any increase in the wages of home care workers is uncertain.

THE DEPARTMENT OF VETERANS AFFAIRS' LONG-TERM CARE PROGRAM

The Department of Veterans Affairs (VA) has its own program of long-term care for qualifying veterans.²⁴ In addition to a retirement pension payable after 20 years of service, there are three classes of benefit for older veterans: the VA Pension, the Aid and Assistance Benefit, and the Housebound Benefit. They have a common service requirement, as well as a common limit on net worth, which in 2021 was \$130,773. The VA's measure of net worth includes a measure of income that consists of salary and related compensation, Social Security benefits, and any other pension benefits. The value of a permanent residence, one vehicle, and certain other real assets that are part of a house are excluded from the limit. A three-year look-back period applies to discourage transfers of assets within three years of an application for benefits.

To qualify for a VA Pension, a veteran must satisfy one of the following four criteria: 1) be age 65 or older; 2) have a total and permanent disability; 3) be a patient in a nursing home receiving skilled nursing care; or 4) receive Social Security Disability Insurance or Supplemental Security Income (SSI).

The value of the VA pension is determined by the difference between the maximum annual pension rate (MAPR), which is set by Congress, and countable income, from which nonreimbursable medical expenses may be deducted. The MAPR depends on the number of a veteran's dependents and his or her marital status. Through November 30, 2021, the MAPR for a single veteran was \$13,931, and for a married vet or a vet with one dependent, \$18,243. The actual payment, which is made monthly, cannot exceed the MAPR, and is reduced to zero if countable income exceeds the MAPR. In 2017, it is estimated that 637,000 vets were receiving the VA pension for at least one month of that year, with a median monthly payment of \$1,087 (Giefer and Loveless 2021).

The Aid and Assistance Benefit is made to veterans who require the services of a paid caregiver for ADLs, are bedridden, live in a nursing home and are physically or mentally incapacitated, or have significantly poor eyesight. The Housebound Benefit is made to veterans who have significant difficulty leaving their residence because of a permanent disability. Each of these benefits has a MAPR, which is higher than the VA benefit. For a single vet with no dependents the combined VA and Aid and Assistance Benefit was \$23,238; for a married vet or a vet with a single dependent, the benefit was \$27,549. The Housebound Benefit falls between the VA benefit and the combined benefit.

RECENT DEVELOPMENTS IN THE PROVISION OF LONG-TERM CARE AT THE STATE LEVEL

In recent years, six states have explored the idea of a program that would provide their residents with some or all the services that Medicaid now provides. Washington State, in fact, has implemented such a program. In 2019, Washington's state legislature passed a program that provides qualifying residents up to \$100 per day in support for up to one year, or a total lifetime maximum of \$36,500 as a bridge to support from Medicaid. Washington's program does not require that residents meet the income and asset tests that Medicaid imposes, which are described in Chapter 2. Medical grounds for eligibility are the same as those that apply with the state's Medicaid program. The program is financed by a payroll tax of 0.58 percent on all employees including the self-employed. In addition to helping residents aged 65 and older, the program reduces the cost to the state of its share of Medicaid expenses that it would otherwise incur. The program is vested and requires a certain period of contributions, among other conditions, before a resident may become eligible. Full program implementation is scheduled to begin in January 2025, with premium

collection beginning in January 2022. Washington State's program will ease both the burden of those of its residents who require either institutional or home-based care, as well as the burden of Medicaid on the state's finances.

The other five states (California, Hawaii, Maine, Michigan, and Minnesota) are still developing their policies. Hawaii, the furthest along of the five, has reached what has been described as the preoperational stage (Cohen et al. 2020). Taken as a group, these states are relatively progressive in the sense that their social policy programs tend to be better developed than those of many other states. It is uncertain whether other states will follow their example.²⁵

Notes

- 1. This chapter draws heavily on Kaiser Family Foundation (2015), as well as the other sources noted. This account is not exhaustive, but rather seeks to highlight the thrust of legislative initiatives to establish and broaden the coverage and the quality of long-term care. Private long-term care insurance, which can pay for either institutional or at-home care is discussed in Chapter 2. As a historical aside, the origin of organized social welfare programs can be found in the pension program established for Civil War veterans in the 1870s. In the 1880s, the administration of the program was moved to Washington's magnificent Pension Building, which is now home to the National Building Museum. Lepore (2018), however, traces the dawn of organized social welfare even further back, to the establishment of support by states of the Confederacy for the widows of Confederate soldiers.
- An estimated 76 percent of caregivers of adults are responsible for one person, with 24 percent caring for two or more (AARP 2020a). The calculation presented in the text assumes that the 24 percent who care for more than one care for exactly two adults.
- 3. AARP (2020a) presents survey results for caregiving in general. AARP (2020b) addresses caregiving for adults aged 50+. The two publications report the same figures for recipients of care aged 50+.
- 4. According to Smith and Feng (2010), roughly 2 percent of the elderly population were housed in either local poorhouses or state psychiatric hospitals prior to the establishment of the SSA. As of 2016, the share of the population aged 65+ who resided in nursing homes is estimated to have been about 2.5 percent. The figures are not strictly comparable,

because the figure for nursing homes does not include elderly persons in long-term psychiatric facilities.

The Poor Relief Act of 1601, passed by the English parliament during the reign of Elizabeth I, provided that local authorities (parishes) were to care for the aged, the blind, and those unable to work in an almshouse or a poor house.

- 5. Private boarding houses tended to reemerge as for-profit nursing homes a sector that continues even today to serve a larger proportion of the poor elderly population than do nonprofit homes (Smith and Feng 2010).
- 6. In his classic work Asylums, Goffman (1961) includes homes for the aged among what he terms "total institutions." Although his examples are mainly drawn from mental hospitals and prisons, his characterization of a total or closed institution: regimentation, tight scheduling, and doing the same thing every day with the same people in the same places, is unsettling in the way it pinpoints key features of nursing homes then and now.
- 7. The six ADLs are 1) bathing, 2) toileting, 3) eating, 4) transferring in and out of bed, 5) being mobile, and 6) dressing and undressing.
- 8. States have the option of offering the following additional home health services: physical therapy, occupational therapy, and speech pathology and audiology services.
- 9. Strictly speaking, moral hazard may not be the most accurate term. It normally refers to the tendency for insurance of a contingency to encourage behavior that makes the contingency more likely. For example, moral hazard occurs with life insurance if insured persons take up hang gliding or engage in other risky behavior. Here it is essentially referring to the possibility that persons who do not really qualify for home care may falsely claim that they do—that is, dishonesty.
- 10. In the 1980s, some states made their Medicaid nursing home eligibility and screening policies more stringent to reduce demand for beds. At least 30 states instituted formal preadmission screening programs for Medicaid nursing home placements in the 1980s to ensure that services were needed; this became mandatory with the adoption of the 1987 Omnibus Budget Reconciliation Act (OBRA) nursing home reform legislation (Harrington et al. 1992).
- 11. These agencies include MACPAC (Medicaid and CHIP Payment and Access Commission), CMS (Centers for Medicare & Medicaid Services), and the NCHS.
- 12. In particular, the 1988 Long-Term Care (LTC) Assistance Act, the 1988 Life-Care LTC Protection Acts, the 1990 Pepper recommendations, and the 1993 Clinton Health Security Act never made it out of Congress. The Community Living Assistance Services and Supports program

- was passed as part of the Affordable Care Act. It was designed to be a publicly administered voluntary program of long-term insurance, but doubts about its financial viability contributed to opposition leading to its repeal in 2013.
- 13. National Center for Health Statistics (2019) appears to have the most comprehensive data on the number of nursing homes by type of ownership, beds, and the survey information discussed in the previous section. The latest years for which data are available are 2015–2016, although data on the number of residents through 2019 is available from other sources. The Nursing Home Data Compendium (2015) covers the 2005–2014 period (see CMS 2015).
- 14. Infection prevention and control deficiencies cited by surveys can include situations where nursing home staff did not regularly use proper hand hygiene or failed to implement preventive measures during an infectious disease outbreak, such as isolating sick residents and using masks and other personal protective equipment to control the spread of infection.
- 15. In 1965, Congress had already passed the Older Americans Act (OAA) in response to concerns about a lack of community social services for older persons. The original legislation established authority for grants to states for community planning and social services, research and development projects, and personnel training in the field of aging. The law also established the Administration on Aging (AOA) to administer the newly created grant programs and to serve as the federal focal point on matters concerning older persons. Today the OAA is considered an important albeit not the only institution for the organization and delivery of social and nutrition services to this group and their caregivers. The OAA also promotes community service employment for poor older Americans; training, research, and demonstration activities in the field of aging; and activities that protect the rights of the vulnerable elderly (Administration for Community Living, n.d.). The act was last reauthorized in 2020.
- 16. Gleckman (2020) argues that a large number of people are in nursing homes because they cannot obtain the home-based services they need from their state, not that they really need the services provided by a nursing home.
- Spousal impoverishment safeguards are explained in Chapter 2, on Medicaid financing.
- 18. The figure for the total number of persons benefiting from optional programs reflects some double-counting, because one person may be receiving benefits under a state program and a Medicaid waiver program.

- 19. The role of family caregivers is uncertain. Many states, under the Home and Community Based Services State Plan Option allow relatives to be hired by the care recipient. However, only 14 states allow a spouse as caregiver, so that usually an adult child would be paid by Medicaid, if at all. These plans are entitlement plans, meaning that any recipient of care is in principle eligible. Another program, the Community First Choice Program, allows persons who need nursing-home levels of care to get that care at home. Some nine states have adopted this program. Two other programs for consumer-directed care—HCBS Medicaid Waivers and Section 1115 Demonstration Waivers-exist, but these give the states the right to restrict the number of recipients and the relationship of the family caregiver as well as the area of the state in which the program is available.
- 20. Personal care aides provide other household assistance and assistance to individuals who want to remain a part of their communities. Home health aides (and in some cases, nursing assistants) may perform certain clinical tasks under the supervision, which may be remote, of a licensed professional.
- 21. The training might be online or in-person and included topics such as safe lifting practices, allergy management, and tracheotomy and gastrointestinal tube care (for nursing aides).
- 22. The median nominal wage per hour for direct care workers increased from \$12.56 to \$13.56 between 2010 and 2020, or by 8 percent. Consumer prices actually increased by 17.2 percent (PHI 2021).
- 23. The aging in place movement has as its principal goal keeping people in their own homes and out of congregate institutions for as long as possible.
- 24. This section is based on A Place for Mom (n.d.), and Department of Veteran Affairs (n.d., 2021).
- 25. The vesting and residency requirements of Washington's program would discourage in-migration of non-state residents seeking to benefit from the new program.

Chapter 2

How Care Is Financed

MEDICAID

Medicaid is a program for poor, or relatively poor, elderly (aged 65+ years) and disabled Americans. It is financed jointly by the federal government and the states from general tax revenues, and in this respect differs radically from Medicare, which is financed by payroll taxes, levies by the federal government on taxpayer incomes exceeding certain thresholds, and the premiums paid by persons aged 65+. This chapter's principal focus is on Medicaid's role in the financing of long-term care for people aged 65+, but although persons benefiting from long-term care services and supports amounted to just 5.5 percent of Medicaid beneficiaries in fiscal year 2019, they accounted for close to one-third of Medicaid expenditure (Rudowitz et al. 2021).

Medicaid provides a guarantee of federal payments to match state expenditure and is open-ended in the sense that it has no preset limit; that is, any person who satisfies the eligibility requirements set by the federal government—or, in the case of certain services, by the states that have opted to provide the services—is covered. With Medicare and Social Security, however, the rules that determine eligibility are the same nationwide. With Medicaid, although states are required to offer certain services to be eligible for federal support, the rules for other services may vary from state to state, even within a state, and are determined by the states themselves. Much of the complexity of Medicaid's financial arrangements stems from its joint ownership by the two levels of government and by the critical role played by means-testing in determining eligibility. Means-testing plays no role in either Medicare or Social Security.²

The share of a state's Medicaid expenditure paid by the federal government varies according to a formula under which the share var-

ies with a state's relative per capita income. The minimum share is 50 percent of combined federal and state expenditure, and the maximum is 77 percent as of 2021.3 The minimum a state receives from the federal government is thus one dollar for each dollar the state itself spends. The share paid by the federal government applies to both the programs that the state is obliged to provide (nursing homes and that part of HCBS that is required under Medicaid's terms, as discussed in Chapter 1), as well as optional HCBS programs. The federal-state sharing arrangement has led some of its critics to argue that the states effectively overspend, on the grounds that the marginal benefit of extra expenditure to a state is less than its total marginal cost. These critics contend that a block grant arrangement—where the federal government pays each state a lump sum—would be preferable, because any expansion of benefits beyond the block grant would be entirely financed by the state.⁴ However, growing income inequality, which is linked to worsening health as well as increasing health costs, have undoubtedly pressured states to spend more on Medicaid.

Medicaid provides special match rates for the Affordable Care Act's Medicaid expansion, and for administration and other services. A state may also receive a higher percentage for certain services or populations. Medicaid also provides disproportionate share hospital payments to hospitals serving many Medicaid and uninsured patients.⁵ The Medicaid expansion that took place under the Affordable Care Act is financed primarily with federal dollars and accounts for a relatively small share of total Medicaid spending. As of April 2021, 12 states continued to opt out of the expansion of Medicaid under the Affordable Care Act, despite the small share of the additional cost they bear.

Notwithstanding Medicaid's cost-sharing arrangements, state spending on Medicaid as a whole amounted to about \$225 billion in FY 2019, which is a sizeable share of state tax revenues. In an effort to reduce the burden that their share of the cost of Medicaid entails, states have reduced the coverage of HCBS programs or reduced expenditure per capita on these programs and the services they are

obliged to provide under Medicaid. Enrollment tends to increase during recessions, when states' revenues are declining and tends not to reverse that increase with economic recovery, which adds to the pressure on states to hold down expenditure per capita.

ELIGIBILITY FOR MEDICAID'S INSTITUTIONAL AND HCBS COVERAGE⁶

The share of the population that is eligible for Medicaid has grown substantially in recent years. Originally, its coverage was restricted to aged, disabled, and blind persons who had qualified for SSI, which was 74 percent of the federal poverty line. The growth in coverage reflects both the inclusion of additional elements of the population and the adoption of less restrictive but certainly not generous means tests. In the 38 states and D.C. that have now opted to participate in the Affordable Care Act's expansion of Medicaid in 2015, virtually all state residents whose incomes do not exceed \$17,775 per year as of 2021 (which was 138 percent of the federal poverty level for a single person) are covered at the state's option.

Medicaid relies on an income test as well as an asset test to determine eligibility for both health care and long-term care.⁷ The income and asset limits included in the original legislation apply to what is known as the Aged, Blind, and Disabled pathway, which covers the original group to be eligible for Medicaid. The original income limit equaled 74 percent of the federal poverty line, but some states have the option to increase the limit up to 100 percent. The income test excludes income from government programs, as well as a small part of earned income.8 As of 2018, the monthly income limit for all 50 states and D.C. ranged from \$528 in Connecticut to \$1,164 in Hawaii for an individual, with a median value of \$750, and from \$696 to \$1,578 for a couple, with a median value of \$1,125. In about half of the states, the limit for a single person fell to between \$700 and \$800 (Musumeci, Chidambaram, and O'Malley Watts 2019).9

The asset limit for an individual ranged from \$1,500 in New Hampshire to \$7,560 in Arkansas, and for a couple, from \$1,500 in New Hampshire to \$11,340 in Arkansas. Most states impose a limit of \$2,000 for individuals and \$3,000 for couples. The limits appear to be low, but they exclude the value of an applicant's primary residence (up to a value of \$560,000, or as much as \$840,000 at the option of the state), one automobile, and personal property and household belongings. These exclusions effectively make the asset test a limit on financial assets, not real assets, for most older Americans. The treatment of retirement plan balances under the asset test varies from state to state: a few states exclude them entirely. Warshawsky and Marchand (2017) estimate that in 2010, some 71 percent of retirement plan assets were countable toward Medicaid asset tests.

For older Americans in need of long-term care, two other pathways to Medicaid eligibility are available that broaden the program's coverage considerably. The first of these is the special income rule, under which 42 states and D.C. have chosen to increase the income limit to three times the standard payment for the SSI, or three times \$794 or \$2,382 per month in 2021. An asset test that for most states is similar to the test for the Aged, Blind, and Disabled pathway also applies. The income limit of the special income rule is considerably higher than the limit applying under the original program and also higher than the limit that may be applied by states that opted for the Affordable Care Act's expansion of coverage, and three of every four Americans aged 65 years and older are residents in states that offer this pathway. 12

The second pathway of eligibility is the medically needy pathway, which provides some coverage for applicants whose medical expenses take up a large share of their income. Thirty-four states have opted for this program, 26 of which have chosen also to apply the special income pathway.¹³ Its income test is based on a monthly income limit that each state sets. Applicants whose income exceeds that limit are not eligible through this pathway unless they are able to show that they have incurred medical expenses that equal or exceed

the difference between their current monthly income and their state's monthly limit over a stipulated period that ranges from one to six months, depending on the state. 14 Monthly income limits for an individual in 2018 ranged from \$100 in Louisiana to \$1,041 in Vermont. The median value for an individual was \$488. Limits for a couple ranged from \$192 in Louisiana to \$1,372 in Illinois. The median for a couple was \$559.

The application of this rule is complex, and an example of how it might be applied may be helpful. Consider an individual applicant with a monthly income of \$2,800 and who is a resident of West Virginia, where the monthly income limit is \$200 and the budgetary period is six months. The applicant would have to incur \$15,600, or $(\$2,800 - \$200) \times 6$ in medical expenditures to be eligible for Medicaid's coverage on additional expenditures for the remainder of the budgetary period. To take another example, suppose an individual has a monthly income of \$6,500 in a state with a monthly income limit of \$500. With a budgetary period of six months, the individual must incur medical expenses of \$36,000, or (6 × \$6,000), before Medicaid would cover excess expenditures for the remainder of the budgetary period. Once that period ends, the whole procedure must begin again. This second example makes clear that the relief the medically needy pathway offers drops significantly with increases in income.

Any American who satisfies the asset test and whose income falls below the limits set by the original Medicaid program or the special income test is effectively insured by Medicaid against the risk of requiring long-term care. The fact that many older Americans can expect to rely on financing from Medicaid should they need longterm care conceivably may reduce the incentive they have to save for their declining years. Nonetheless, nursing home residents are expected to contribute most of their income to defraying the costs incurred on their behalf before Medicaid kicks in. 15 They are allowed only a small personal allowance. If only a little money is remaining after the nursing home takes its share, the sharing rule may be inflicting hardship. How much discretionary income a nursing home resident would require over and above what is needed to cover food, other basic living expenses, and the cost of care is an important issue. In the case of a nursing home resident with a spouse, rules are in place to avoid the spouse's impoverishment because of large long-term-care expenditures on behalf of his or her chronically ill spouse.

The financial arrangements for Medicaid's coverage of long-term care differ substantially from those of Medicare's coverage of short-term illness or disability. All Americans aged 65 or older who qualify for Social Security and pay their share of the payroll taxes that finance it and Medicare are eligible for Medicare. Medicare Part A, which covers hospitalization, is normally free of charge. Part B, which covers the fees of doctors and other health care providers, requires a premium that varies with income. Part C, which is known as Medicare Advantage, provides coverage through HMOs and several other vehicles, and can substitute for Medicare Parts A and B. Part D is a prescription drug program provided by private plans. It includes a catastrophic coverage limit. Mackenzie (2020) offers a summary description of the whole program.

Unlike the risk of exposure to health care costs of Americans aged 65 or older, the degree of exposure to long-term care risk depends on income and the state of residence. Older Americans whose incomes are below the SSI limit and who satisfy the asset test are covered. In addition, older Americans are covered if they are a resident in one of the 42 states or D.C. that have adopted the special income pathway, have an income less than three times the SSI limit, and satisfy the asset test. In the remaining eight states, the medically needy pathway also may provide relief to some, but the asset test may require a substantial spend-down.¹⁶

Of the 43 states (including D.C. among the 43) that do grant eligibility through the special income pathway, only 26 also offer eligibility through the medically needy pathway. Older Americans in the other 17 states with relatively high incomes, and without private long-term care insurance, must rely entirely on their own resources to pay for long-term care (Table 2.1).

Table 2.1	Number of States with the Special Income Pathway and/or
	the Medically Needy Pathwaya

States offering both pathways	26
States offering only the special income pathway	17
States offering the medically needy pathway	8
Total, including D.C.	51

^a All states and D.C. offer at least one of these two pathways.

SOURCE: Musumeci, Chidambaram, and O'Malley Watts (2019) and author's calculations.

Relying on Medicaid rather than taking out private long-term care insurance makes some sense for older Americans with modest incomes. Note, however, that while relying on private long-term care insurance requires the payment of premiums typically starting some and often many years before need, private insurance normally pays all or at least much of the expenses of long-term care once care begins, and long-term care policyholders would have more choice regarding care either in a congregate setting or at home than they typically would under Medicaid. Older people with private long-term care insurance would be much more likely to afford residing at an ALF than the typical person without that insurance.

If older Americans cannot qualify for long-term care benefits under Medicaid and lack private long-term care insurance, what expenses would they face? A 2017 study finds that the probability that one or both members of a healthy 65-year-old couple will move to a nursing home at some point in their remaining lifetime is 78 percent; the probability of visits by a home health aide is 63 percent; and the probability of residing in an assisted living facility is 29 percent (Crook and Sutedja 2017).¹⁷ Despite the likelihood that older Americans will need some kind of long-term care, the median duration of a nursing home stay was estimated to be only 9 months. The median duration of home health visits and residence in ALFs was 14 months. A more recent study finds that the probability of a 65-year-old in 2020 developing a disability severe enough to require long-term care is 56 percent (Favreault and Dey 2021).

With the median annual cost of a nursing home stay in 2020 at about \$93,000, the cost of long stays becomes prohibitive.18 The chances that a nursing home stay lasts for more than one year has been estimated at 32 percent for at least one member of a healthy 65-year-old couple; the probability of a stay of more than three years has been estimated at 8 percent (Crook and Sutedja 2017). The odds that an older American who does not qualify for Medicaid will incur substantial expenses for long-term care are not negligible, but neither are they large. Comparing the contingencies covered by auto and long-term care insurance, it is highly unlikely that Americans can pass through life without having to deal with an auto accident of some degree of seriousness, and consequently most people have at least an intuitive understanding of the benefits of auto insurance.¹⁹ The probability of requiring long-term care is lower than that of a serious car crash or car theft, however. As a result, wishful thinking may lead many people to underestimate their exposure to the latter contingency, or to expect that family members will be able to care for them, without considering the emotional and financial burden that home care can involve.

The older Americans who are most exposed to the risk of longterm care expenses are those who live in the 17 states that do not offer eligibility through the medically needy pathway, and whose incomes exceed the limit of three times SSI. Residents in these states who are 65 years or older account for about 30 percent of the national total. Assuming, perhaps conservatively, that no more than 50 percent of the elder population fails this test (i.e., that one in two has income more than three times the SSI limit), it would amount to about 14 percent of the population age 65 years and older. The asset test, however, also must be met. If it cannot, then candidates for long-term care must spend down those of their assets that are deemed to be countable for the purposes of the asset test or protect them by establishing a trust.²⁰

ACTUARIAL AND FINANCIAL ASPECTS OF CONTINUING CARE RETIREMENT FACILITIES

Viewed from the economic point of view, the basic feature of a CCRC is that it is a risk-sharing arrangement, one that entails obvious risks for both parties.²¹ Prospective residents in CCRCs, as well as CCRCs themselves, confront longevity risk—the unpredictability of lifespans. Both parties may also confront investment risk, the risk of an uncertain rate of return on their reserves. The degree of both longevity and investment risk depends on the nature of the contract between the two parties.

Longevity risk is a basic risk for the CCRC (i.e., the owner or owners) and for prospective residents, who have greater life expectancies than the general population. In part, this reflects the fact that CCRC residents are wealthier than the general population, and there is a positive relationship between wealth and longevity.²² A further influence is adverse selection—persons interested in joining a CCRC are likely to believe that they will live there for many years, particularly if the move requires that they sell a home in which they have lived for many years, which is not an easy step for many older people to make. It is also argued that the community aspect of living in a CCRC promotes longevity.

In addition to longevity (or mortality) risk, there is also morbidity risk—the risk that a resident may need to move to the assisted living or nursing care component of the community, which increases the costs for the CCRC. It is important to distinguish between the risk of a temporary transfer to the nursing home component and a permanent transfer, that is, one that lasts until the death of the resident.

As Chapter 1 explains, the typical CCRC contract involves a large entry fee with continuing monthly payments until the death of the resident, or both residents if a couple. A prospective resident may have a choice between several combinations of entry fee and monthly payment. With the largest entry fee, there is no increase in the monthly

fee in the event of a transfer to the nursing component. Residents may also be offered a trade-off, where in return for a smaller entry fee, they accept the risk that their monthly fee may increase in the event of a transfer. Not all CCRCs offer the intermediate step of assisted living between independent living and the nursing home component, and some limit the number of days that may be spent in nursing at an unchanged monthly fee. A rental arrangement is also possible, where there is no sizeable up-front entry fee.²³

With the large entry fee, the CCRC is taking on the risk that its earnings and the fee itself will not cover the expenses of a very long-lived resident, or one who will require many years of nursing-level care. Contracts typically provide for the partial or total reimbursement of the entry fee if the new resident dies within some stipulated period after entering the community. However, once that period has elapsed, no refund is possible. In addition, contracts will typically provide for periodic adjustments in monthly fees to reflect increases in the cost of providing a given level of service.

Without having precise numbers for entry fees and monthly payments, it is impossible to draw hard and fast conclusions about the advantages and disadvantages of different contracts for prospective residents and owners. If, however, we compare two basic versions of a CCRC contract—one with a large entry fee and one with a smaller entry fee, and correspondingly smaller and larger monthly fees—we may draw some qualitative conclusions. The large entry fee contract may seem more attractive to residents who worry about their ability to afford an increase in monthly fees in the event of serious illness or who have enough of a nest egg that paying the larger entry fee will not unduly deplete their reserves. Residents choosing a smaller entry fee also incur the risk that the return on their reserves may be disappointing.

From the CCRC's perspective, the contract with the larger entry fee may look attractive if it thinks that residents overestimate their own longevity. There is also the advantage of getting the money up front and avoiding the complications that arise if a resident cannot pay

the monthly fee.²⁴ However, in agreeing to the contract with the large entry fee, a CCRC is more exposed to the consequence of an extended stay in the nursing home component of the facility.²⁵ Another potential risk for the CCRC is a decline in new entrants, which some connected with the industry have noted can occur with declines in the market for residential real estate, because the sale of a household's principal residence often provides the money for the entrance fee.

Regulatory Issues

As of 2018, only 38 states had any kind of regulatory framework for CCRCs (Breeding 2018), the same number as GAO (2010) reported. The remaining states and D.C. had none. Broadly speaking there are two distinct aspects to the regulation of CCRCs: the regulation of their role as providers of health care, which pertains mainly to their assisted living and nursing home components, and their financial regulation. All 38 states regulate the provision of health care in tandem with the federal government (as do all the states and D.C.). In Maryland, for example, the CCRC regulatory function is the responsibility of the Department of Aging. The department's website notes that only part of the contractual arrangements of CCRCs are subject to regulatory law. On its website, the department "urges anyone who is considering moving into a CCRC to consult with an attorney and a financial advisor familiar with these types of agreements before signing any documents."²⁶ This is certainly good advice, but one wonders if it is enough.

It appears that the 38 states that regulate CCRCs do require that they provide basic financial accounting statements, such as a statement of assets and financial liabilities, as well as income and expenses. The former would include a valuation of the CCRC's real property with an estimate of depreciation, as well as financial assets and any debt. A basic income statement would include among its elements on the income side monthly fees paid by residents, in addition to payments for other amenities like cable, internet, and dining subscriptions.

Breeding (2018) notes that the mandatory requirements and degree of oversight among the 38 states can vary substantially. North Carolina requires annual audited financial statements with disclosure statements. Operating reserves must be maintained at 50 percent or more of forecasted operating revenue for the next 12 months, although this ratio may drop to 25 percent for CCRCs with occupancy rates of 90 percent or higher. Breeding also notes that no CCRC in North Carolina has gone bankrupt to date. Some states are said to have less stringent reporting requirements, and apparently some have more.

The GAO (2010) report was based on a study of CCRCs in eight of the country's most populous states.²⁷ It found that only three of them required actuarial valuations.²⁸ A financial statement of assets and liabilities can make a CCRC appear to be in good financial condition when it is seriously undercapitalized. A periodic actuarial valuation of a CCRC's contingent liabilities is needed for a comprehensive evaluation of its true financial condition. Quite apart from these actuarial issues, there have also been some cases of misappropriation of financial assets.

PRIVATE LONG-TERM CARE INSURANCE

The market for private long-term care insurance has seen better days.²⁹ Its origins date from the early 1990s, and by 2000 about 100 insurance companies were in the market. However, their numbers began to shrink subsequently (Cohen 2019). According to Cohen (2019), the number of active claims in 2019 was about 300,000, and there were about seven million policies.

In its current form, a policy for private long-term care insurance stipulates a maximum lifetime benefit payout and a typically maximum payout period as well as a maximum daily payout, which may vary depending on whether the policyholder lives in a congregate setting like a private nursing home or an ALF or lives at home and receives trained care. Home caregivers are typically provided through

an agency that the insurance company has vetted. The insurance company pays the agency on behalf of the policyholder and the agency pays the caregiver and is responsible for withholding payroll and income taxes. Caregivers who perform basic tasks typically receive a modest wage.³⁰ Some policies may also pay family caregivers a modest stipend.

Long-term care insurance issuers are subject to regulation at the state level. States require that once a policy has been issued it may not be revoked unless the policyholder is unable to pay the premium, which is normally due annually. The premium is supposed to be set so that, taking account of the reserves that will accrue on premiums, it will not need to be adjusted, apart from changes that the regulating state deems are necessary given changes in underwriting costs. In their adjudication of requests for premium increases from insurance companies, states are expected to strike a balance between the interests of the policy issuer and policyholders.

Premiums increase with the age at which the insurance is contracted. The younger the age, the longer will be the expected period that elapses before a claim is activated, because the need for longterm care insurance increases with age, and because with an increase in the interval between the start of coverage and the activation of a claim the longer the period over which reserves can accumulate. The level at which a premium is set is also a function of lapse rates, and the younger the age at which coverage begins, the greater the chances that it will be allowed to lapse.

To trigger a claim, a policyholder must demonstrate to the insurer that he or she is unable to carry out a number of the six ADLsusually two—or display a significant degree of cognitive decline.³¹ This assessment is necessary to deal with the problems of moral hazard, which we touched on in Chapter 1 in the discussion of publicly provided long-term care, and adverse selection. Moral hazard can crop up with long-term care insurance when the policyholder exaggerates or falsifies a claim of inability to perform ADLs. Adverse selection, the tendency for an insurance arrangement to attract clients who are

more likely than the average person to need the insurance, is also a feature of this market and contributes to the gap observed between the premium for a group policy and an individual policy, the former being less susceptible to adverse selection than the latter.

There are several reasons for the decline in this market. Insurers underestimated the longevity of claimants and overestimated lapse rates, and did not foresee the rate at which the cost of care increased. The general decline in interest rates that has taken place over the past 20 years has required an increased buildup in reserves to offset a given liability. These developments have increased break-even premia and reduced demand. An NAIC study (Nordman 2016) found that the average income of new policyholders had increased substantially since the early 2000s. The American Association for Long-Term Care Insurance (2021) reports that in 2021, a select policy with a lifetime maximum of \$165,000 would cost a 60-year-old woman \$1,900 and a 60-year-old man \$1,175. If the maximum benefit of the policy increases by 3 percent per year, premiums would be \$4,300 and \$3,525, respectively. These figures are high enough to deter many middle-income households from acquiring a policy given the uncertainty surrounding the eventual need for long-term care and the age at which care would be needed.

Price is not the only influence on demand for private long-term care insurance, however. Both Medicaid and private long-term care insurance provide protection against catastrophic loss entailed by a long period of care, although Medicaid is subject to the stringent means-testing previously described, and private long-term care insurance is not. Self-insurance against such a contingency is possible only for the rich. Consequently, private long-term care insurance and the publicly provided benefit may compete with one another.³² But private insurance has clear advantages over Medicaid, most notably, the protection of assets and income from means-testing, the choice of facility if institutionalized care is required, and perhaps also the quality of care at home. However, ignorance about the costs of Medicaid also plays a role. Many Americans believe that Medicare, not Medicaid,

covers long-term care, and many are unaware of the stringent eligibility requirements that Medicaid imposes, especially on those who have not protected their assets by establishing a trust.³³ Greater awareness of the advantage of private insurance as well as a greater awareness of the real risk of one day needing long-term care would undoubtedly increase demand for it. However, it is easy for people in their fifties and sixties to downplay that risk because of its uncertain timing and the fact that the risk increases with age. A further inhibiting influence on demand may be the unpredictability of premium increases.

Private long-term care insurance could cover more U.S. households than it does. One promising development has been the emergence of so-called combination products, such as the life care annuity, whose payout increases if the policyholder demonstrates a need for long-term care. The product reduces underwriting costs because its potential clientele have shorter life expectancies than traditional annuitants. The market might also grow if private insurance could take the form of a "front-end" product that covered the first two years or so of need before Medicaid kicked in. This would reduce its cost. However, the cost of traditional private insurance policies and the existence of a public option are obstacles to the development of a larger market for them.

UNPAID CAREGIVERS

As emphasized in Chapter 1, the number of older Americans in need of assistance with one or more of the ADLs, living at home, and being helped by a relative or in some cases someone from outside the family dwarfs the number living in congregate settings. These caregivers should not be confused with either the health care workers who come to the homes of elderly persons in need of assistance with ADLs or instrumental ADLs or those who provide the more specialized services provided under Medicaid's HCBS programs.

As noted, above, Medicaid does pay some of the relatives and others who provide basic care, but the terms vary considerably from state to state, and it is extremely difficult to know how many of the huge number of the nation's caregivers receive any money at all. In fact, the AARP studies of home care already cited (AARP 2020a,b) effectively assumes that all home caregivers are unpaid.

That there are so many unpaid caregivers raises difficult economic, financial, political, administrative, and moral or social issues. These issues intertwine and overlap, and it is not easy to pull them apart to analyze. However, it is worth trying to do so. Our discussion begins with the moral aspects of the issue. Unpaid caregivers shoulder an obvious economic and psychological burden. Studies by AARP have repeatedly found that the great majority of caregivers also work for pay outside the home and come from all age classes. The time they spend as caregivers reduces the hours that those of them who are potential labor force participants can work for pay outside the home, and the hours they can devote to other productive activities within the home, such as meal preparation and supervising homework—or simply relaxing, which of course is also true of caregivers who have retired themselves.

This economic cost caregivers and their families bear raises the question of whether society at large—that is, the taxpayer—should compensate them for looking after their older and infirm relations.³⁴ Doing so would of course require an increase in taxes and/or a reduction in other public expenditure programs, and many observers might argue that the benefit of the activity of caregivers does not accrue to society at large but to the families of the people being cared for. But the same argument might be advanced to oppose taxes that finance education or child care for low-income families, if one argues that these taxes benefit only families with school-aged children, not older families.³⁵ In addition, caring for persons aged 65 or older must reduce the demand for beds at nursing homes, which does reduce the general tax burden. Caregiving also reduces the time that givers can devote to their own children, and to furthering their own education.

This may impose a cost on society. More generally, if we accept that society at large has a moral obligation to provide some support for the disabled elderly, then we also effectively assume a responsibility of not allowing an undue share of that burden to fall on their caregivers. This is a moral or an ethical judgment and not a statement that can be refuted by evidence. By this point the reader will have realized that it is a judgment the author shares.

Caregiving has an obvious economic cost, what economists call its opportunity cost—caregivers could be doing something else productive with their time. AARP has published a regular series of studies that address the issue of what it has aptly termed "valuing the invaluable." These studies make calculations of the economic cost of the time spent by caregivers, which can serve as a basis for calculating what a public policy of remunerating them would cost the public purse. The latest of the AARP studies is based on four different sources, with various estimates of the number of caregivers, the average number of hours they work, and the average wage that should apply to their work (Reinhard et al. 2019). The AARP's weighted analysis of these studies yields a total wage bill or cost of \$370 billion, which is about 2 percent of 2017 nominal GDP, the year for which the data was collected.36

In macroeconomic terms this is not a tiny sum. It would require a substantial increase in the national tax burden to finance it—at the federal level, if the benefit were paid by the federal government, which would mark a shift away from the traditional sharing of costs by Medicaid programs, the tax burden would have to increase by over 10 percent of all federal tax revenues, including the payroll taxes that finance Social Security.

Any increase in taxation is politically contentious. The nature of the expenditure this tax increase would be financing also raises administrative issues. In particular, the issue of monitoring the work of caregivers arises. Should caregivers be required to submit a verifiable accounting of the number of hours they put in each week? Although requiring a submission at periodic but not too frequent intervals might

not impose a huge paperwork burden on the caregiver, verifying its accuracy would undoubtedly impose an unrealistic burden on the monitoring agency of the government concerned. One approach that would avoid excessive compliance burdens on caregivers or monitoring costs for government might be to require simply demonstrating that an older family member was in obvious need, perhaps once per year, and was being cared for. No accounting of the number of hours of care provided would be required. Instead, some stipulated number of hours of care might be assumed. This approach would amount to a sort of presumptive assessment of the burden of care, at least in terms of hours, and could make sense from an administrative point of view.

This chapter has covered a broad range of complicated issues. Its basic theme is that the current system of long-term care is both excessively complicated and underfinanced. Its complexity means that its basic features are not well understood either by many persons who will soon need long-term care or by the taxpayer. The lack of financing for care at home is inefficient to the extent that it results in institutionalization that is more costly than care provided at home (Kaye, LaPlante, and Harrington 2009). The lack of financing may be judged as inequitable to the extent that society adopts an ethical standard that older infirm persons deserve an adequate standard of care, whether at home or in an institution.

Notes

- 1. This section draws largely on Rudowitz et al. (2021).
- 2. To be eligible for Medicare, a person must have reached the age of 65 and have worked and contributed to Social Security for 40 quarters, regardless of his or her state of residence. To draw a retirement pension, a person must have reached the age of 62. The retirement pension increases with the number of years worked and average wages as calculated by Social Security (actually the 35 years with the highest wages, which are indexed to the economy-wide wage rate for the calculation). The benefit, once elected, is indexed to consumer prices and also increases with the age at which it is elected up to age 70. Some of Social Security's rules are quite complex but they do not vary from state to state.

- 3. Puerto Rico and other U.S. territories are all subject to a limit on the federal government's share of 50 percent—so there is no adjustment for per capita income—as well as a dollar limit. If the dollar limit is exceeded, the federal government's share automatically drops below 50 percent.
- 4. A block grant arrangement has its own drawbacks. The amount of the grant would have to be adjusted regularly just to keep up with inflation. In addition, even an inflation-adjusted grant would have to be adjusted to keep up with population growth, which would vary from state to state. Veghte and Bradley (2017) discuss the drawbacks of block grants and per capita block grants.
- 5. Disproportionate share hospitals (DSH) serve many Medicaid and lowincome uninsured patients. States have considerable discretion in determining the payments to each DSH hospital, and federal DSH funds are capped at both the state and the facility level. Payments to DSH hospitals were about 3 percent of Medicaid's total budget in FY 2019.
- 6. This discussion draws on Mackenzie (2020), which in turn draws on Musumeci, Chidambaram, and O'Malley Watts (2019). It also draws on Skračić, Bond, and Doonan (2020).
- 7. A certain amount of income is disregarded in applying the income test, which varies from state to state but is usually small.
- 8. The limits of the income and asset tests and other limits included in this and the next two paragraphs are for the year 2018 and come from Musumeci, Chidambaram, and O'Malley Watts (2019). Higher disregards apply to boarding homes and shared living (i.e., noninstitutional) arrangements.
- 9. Eight states have elected an option allowing them to use their own income criteria provided these are no more restrictive than what they had in place in 1972.
- 10. Arizona has no limit on asset holdings for either individuals or couples.
- 11. The limit is 250 percent of SSI in Delaware. In Missouri, it varies by program. The eight states that have not opted for the special income rule are California, Hawaii, Illinois, Montana, Nebraska, New York, North Carolina, and North Dakota. These states have all opted to participate in the medically needy pathway described next. Massachusetts does not apply the special income rule regarding institutions (e.g., nursing homes) but does apply it for HCBS.
- 12. In about half of the states, individuals whose income exceeds 300 percent of the SSI can still be eligible for support from Medicaid if they establish what is known as a Miller trust, and they administer through it the income that exceeds 300 percent of the SSI. See Musumeci, Chidambaram, and O'Malley Watts (2019) for additional discussion.
- 13. All 34 states apply this pathway for pregnant women and children; 32

- states apply it for seniors and the disabled (Texas and Tennessee being the exceptions); and 26 states apply it for low-income parents.
- 14. Eleven states have opted for a budgetary period of one month, and 13 states have opted for six months. In several other states, the limit depends on whether the applicant plans to remain in the community (Musumeci, Chidambaram, and O'Malley Watts 2019, Appendix Table 3).
- 15. Recipients of community-based services can retain much more of their income because they will be responsible for much more of the basic costs of living, such as lodging, food and clothing.
- In Massachusetts, as noted, the special income rule covers HCBS but not institutional care.
- 17. The probabilities add up to more than 100 percent because a couple can experience not just one but two or all three of these outcomes.
- 18. Genworth (2020) estimates that the median national cost of a semi-private room in a nursing home in 2020 was about \$93,000.
- 19. Belbase, Cen, and Munnell (2021) find that about one in five 65-year-old Americans will not require long-term care at any level of intensity, while about one-quarter will require at least a moderate to high level of care for some years.
- For a general discussion of estate planning and the role it can play in preserving a family's assets when long-term care becomes necessary, see Correia, Sayre, and Allen (2017).
- The classic study of CCRC finances and actuarial issues is Winklevoss and Powell (1984). The most comprehensive recent study of CCRC regulatory issues is GAO (2010).
- 22. Wilkinson and Pickett (2011) present evidence that, even within the Washington, D.C., area, life expectancy rises enormously as one travels from one end of the Red Line of the Metro system to the other.
- 23. Some observers argue that smaller entry fees are almost always more desirable because most CCRCs do not include in their pricing the benefit of the interest earnings on the entry fee (or the debt service foregone that would otherwise be required without the entry fee). Moreover, entry fees are at-risk investments if the debt a CCRC has incurred is senior to entry fees, which means that the creditors' claims take precedence over the claims of residents. Because the smallest entry fee is no entry fee, some observers argue that pure rental communities are growing in popularity as trust in the entry fee model for financing aging is eroding.
- 24. In one CCRC with which the author is familiar, there are charitable drives to raise funds for residents who are no longer able to afford their monthly fees. Both better-off residents and outsiders can contribute to these charities. In another there is a tacit understanding that no resident who becomes unable to pay the monthly charge will be asked to leave.

- 25. To reduce the risk it accepts in taking on a new entrant, a CCRC will require a physical examination as well as evidence of a new entrant's ability to pay the CCRC's monthly fees.
- 26. https://aging.maryland.gov/Pages/continuing-care-retirement-commu nities.aspx (accessed July 24, 2022).
- 27. The eight states are California, Florida, Illinois, New York, Ohio, Pennsylvania, Texas, and Wisconsin.
- 28. California, New York, and Texas require periodic actuarial studies, but only for those CCRCs that offer contracts that incur long-term liabilities by guaranteeing health care services over the long term.
- 29. This section of the chapter draws on a comprehensive study by Nordman (2016) under the auspices of the National Association of Insurance Commissioners (NAIC) and the Center for Insurance Policy and Research as well as the other sources cited.
- 30. One established agency in the Washington, D.C., area pays caregivers who assist its clients with difficulty performing ADLs or with significant cognitive decline an hourly wage of \$12.00-\$13.00.
- 31. A standard test of cognitive ability is demonstrating the ability to count backwards from 100 by seven.
- 32. Brown and Finkelstein (2009) argue that Medicaid has substantially reduced the demand for private insurance.
- 33. The survey previously noted found that more than 4 in 10 of the general population surveyed, who were aged 50+ years either thought that a public program would pay for most of the cost of six months of longterm care or expressed ignorance (Life Plans 2017). However, the survey also reports a growing awareness that public programs will not step up to the plate.
- 34. Perhaps it is needless to point out that the same issue arises with caring for young seriously disabled family members, although they number far less than older persons in need of care.
- 35. Education economists generally agree, however, that schooling at the primary and perhaps the secondary levels has social benefits that go beyond the benefits enjoyed by the students and their families. The benefits at the tertiary level are considered to accrue mainly to the students and their families.
- 36. The study is based on an estimated nationwide hourly wage of \$13.81. The hourly wage estimates by state ranged from \$10.57 in Louisiana to \$18.01 in Alaska. The figure for the number of caregivers is 41 million, which is based on a weighted average of the survey estimates adjusted to match their demographic composition to eliminate over or undersampling. The calculations assume a 16-hour average work week and a work year of 52 weeks. For further discussion, see Reinhard et al. (2019).

Chapter 3

A Comparative Analysis

Just as health care systems differ enormously across countries, even across the great majority of high-income countries with universal health care, the way in which countries provide long-term care services and supports differs greatly as well. To use Veghte's (2021) terminology, the model that countries follow can be classified as social insurance, universal comprehensive coverage, residual coverage, or a hybrid.

Social insurance is usually financed by payroll taxation, like Social Security in the United States, and its benefits may be limited to contributing workers and their families. In practice, however, coverage is near universal. Universal comprehensive coverage applies to the whole population of a country and is normally financed by the central government's budget. Residual systems are means-tested, and as the term suggests, hybrid systems combine elements of one or more of the first three. Germany and Holland are prime examples of the social insurance model, while the Nordic countries are exemplars of universal comprehensive coverage. The United States and the United Kingdom are both means-tested systems. Veghte (2021) classifies France's system as a hybrid because it is financed by both general revenues and payroll taxes, and because, although its coverage is universal and is not means-tested as such, benefits decline as income increases. Even social insurance systems may have some benefits or services that are means-tested, and coverage may be limited in other ways. For example, room and board in long-term care institutions may not be covered.

A social insurance system can be pay-as-you-go or funded, or somewhere in between. The typical system is pay-as-you-go. With a fully funded system, the contributions of a given age cohort of workers are expected to finance their long-term care needs. When the system is first introduced, it builds up a surplus, at least on paper, because

at its inception contributions far outstrip expenditure. Medicaid is not a funded system. The supplementary system established in Washington State in 2019 and described in Chapter 1 is a funded system.

The role of private long-term care insurance varies across countries, but typically it is a niche market if it exists at all. As Chapter 2 explains, the role of the private long-term care insurance market in the United States is quite limited: as of 2019, outstanding policies amounted to about 7 million, or about 5 percent of the country's labor force, and about 7 percent of the aged 55+ population. Its role in other high-income countries varies. In Germany, some 12 percent of the population has private long-term care insurance, mainly because the federal government puts caps on the amount of support it pays to persons in need of long-term care. However, the government subsidizes the purchase of long-term care insurance to some extent.²

Typically, publicly provided long-term care begins with institutional care before extending its reach to home care. This was the case with the Nordic countries, which were the pioneers in universal long-term care. There has been a general tendency across countries—including in the United States, as Chapter 2 notes—to increase the role of care provided in the home while reducing reliance on institutionalized care. With well-designed policies, this development should increase the welfare of care recipients and reduce the burden of long-term care on public finances.

SOME CONSEQUENCES OF THE MEANS-TESTED APPROACH

A comprehensive study from the OECD (Hashiguchi and Nozal 2020) on the cost to households of long-term care and notably of its impact on the risk of declining into what the study calls relative poverty contrasts the workings of long-term care provision in regions or entire OECD member countries, including two U.S. states, Cali-

fornia and Illinois, as well as England (Scotland, Northern Ireland, and Wales are excluded), France, Germany, and Japan. The study estimates the share of the elderly population that would experience relative poverty, defined as a level of income below the populationwide median, with and without the country or region's long-term care regime, for differing levels of long-term care needs. The United States does not fare well in this international comparison. In both California and Illinois, Medicaid does not reduce the share of the elderly experiencing relative poverty.³

In England, social care is both income- and assets-tested. Minimum allowances guarantee that a share of care recipients' incomes is protected, and care recipients with assets less than £14,250, or about \$19,000, as of December 31, 2021, are eligible for public support covering 100 percent of the costs of care. Conversely, older people with assets worth over £23,250 receive no public support. Apart from the minimum income allowance, they must cover the full costs of care (Hashiguchi and Nozal 2020). However, and unlike the U.S. case, the provision of public assistance keeps most older people in need of long-term care from falling into relative poverty.

The OECD study also compares the share of elderly people already living in relative poverty by the study's definition who do not need benefits under their country's system with the share that would decline into relative poverty if they needed long-term care at home, even with public benefits. Again, the United States fares poorly by comparison with most other countries, including England.

One basic, if obvious, lesson that may be drawn from these analyses is that the impact of means-testing on disposable income after taking account of the out-of-pocket costs that long-term care support does not cover is that it depends on how stringently the income and asset limits are set, and on their design. The OECD study does, however, argue that countries with comprehensive income and asset means-testing generally do less well at protecting the elderly population at risk for needing long-term care from poverty.

Another notable international comparative statistic is the share of GDP that the major industrial countries (the G-7) spend on long-term care (see Table 3.1). The share of expenditures by Medicaid on both residential and HCBS care was 0.9 percent in the United States in FY 2019, which is lower than the other six countries except Italy (for which the reported year was 2017). This is partly related to the comparatively young demography of the United States as well as the limits imposed by Medicaid on the share of the elderly population that qualifies for public support.

Table 3.1 Long-Term Care Expenditure as a Percentage of GDP, G-7 Countries

	Health component	Social component	Total
Canada	1.3	=	1.3
France	1.3	0.6	1.9
Germany	1.5	0.0	1.5
Italy	0.7	=	0.7
Japan	1.8	_	1.8
United Kingdom ^a	1.2	0.3	1.4
United States	_	=	0.9

^a Numbers do not sum due to rounding.

SOURCE: OECD (2019) for all countries except the United States. Author's estimate for the United States based on Rudowitz et al. (2021). The U.S. figure is for fiscal year (FY) 2019. The values for the other countries are for 2017. "Health component" refers to institutional care and care related to the activities of daily living (ADLs), whereas "Social component" refers to care related to the instrumental ADLs.

All the major industrial countries are parsimonious compared to the Nordic countries and the Netherlands, all of which have universal coverage. Norway, Denmark, and Holland all spent more than 3 percent of GDP on long-term care in 2017, and Sweden and Finland spent more than 2 percent.

HOW CANADA AND GERMANY ADDRESS THE CHALLENGE OF LONG-TERM CARE

The long-term care systems in Canada and Germany, which are profiled below, are undoubtedly different in important respects from the U.S. system. Of the two, Canada's is the more similar. They are both more expensive than the U.S. system, and reforms to the U.S. system that make it more like either of them would require a significant increase in public financing. That said, neither the Canadian nor the German system is nearly as expensive as the systems in Holland and Scandinavia. Given the probably insurmountable political difficulties that would be encountered in adopting a version of these latter systems, it makes sense to consider the less expensive systems. Moreover, the systems in both countries, especially Germany, have significant advantages over the U.S. system in the way they provide or finance long-term care to their older citizens.

Canada

The Canadian long-term care system does not fit neatly into one of the three classes of universal coverage, social insurance, or meanstested. Coverage is not universal or near-universal, but care is not means-tested either. These features mean that in some respects the Canadian system is closer to the U.S. system than it is to the systems in most European countries, except the United Kingdom. As was the case in the United States until the passage of the Social Security Act in 1935, long-term care in Canada bore the vestiges of the Elizabethan Poor Laws following the end of World War II, which also saw the establishment of a chain of hospitals for war veterans. In Ontario, Canada's largest province, the Homes for the Aged Act passed in 1949 ushered in the entry of the government into the provision of long-term care.4

The public sector's role in the provision of long-term care in Canada and in its financing are in some ways similar to those in the United States, but in others quite different. In Canada, the provision of both health care and long-term care is primarily in the provincial domain. The provinces and the federal government share in financing institutional care, but there is no long-established quantitative rule to determine the relative share that the federal government pays in any province, as is the case with Medicaid, and the provinces, unlike their U.S. state counterparts, pay for the lion's share of care.

The beginning of a major role for the provinces in long-term care was in the mid-1960s, about the same time that the provincial governments under the aegis of the federal government implemented Canada's version of Medicare, which unlike the U.S. system, covers all Canadians, regardless of age. Like the United States, Canada has both public and privately run nursing homes (called long-term care facilities or care homes), but eligibility for residence is not subject to means-testing of either assets or income.⁵ The provinces set rates at both classes of institution. However, the care provided by long-term care homes is by no means free. Rather, residents or their families pay a monthly rate that depends on the degree of privacy of the accommodation provided. In Ontario in 2019, these rates were about \$1,487 for what is described as basic accommodation; \$1,784 for semiprivate; and \$2,112 for private (Picard 2021).6 Similar rates are said to be charged in other provinces. These costs could pose a heavy financial burden on some middle-income Canadian families, although they are heavily subsidized by the provinces. The total cost of operating Canada's nursing homes is more than double what residents and their families pay.

In 2019, nursing homes had some 190,000 residents. Scaling that figure up by a factor of nine—Canada's population is about one-ninth that of the United States—gives a figure of 1.7 million, compared to the U.S. figure of around 1.3 million. About 6 in 10 residents in long-term care homes suffer from some form of dementia, a share similar to that in U.S. nursing homes, and like U.S. nursing homes, a staff of both skilled and unskilled caregivers is needed at all hours. An additional 170,000 Canadians aged 65+ live in other congregate set-

tings, for a total of about 360,000 residents in all congregate settings, which is about 7 percent of Canada's aged 65+ population, compared to about 4 percent in the United States. The cost of these congregate facilities—retirement communities, ALFs, or the Canadian equivalent of CCRCs—is not subsidized.7

In addition to the lack of means-testing, another distinguishing feature of the Canadian system is the less important role now played by home-based care in Canada compared with the United States. Despite the costs that residents or their families incur for care in longterm care homes, there is a chronic shortage of beds and hence a waitlist for prospective residents. As of June 2017, Ontario—with a population of about 15 million, or less than 5 percent of the population of the United States—was reporting a wait-list of 32,000 beds. The list continues to grow at an annual rate of 15 percent (Canadian Association for Long Term Care 2020). Scaling Ontario's wait-list figure up to the U.S. population would result in a wait-list equal to about onehalf of the number of beds in the United States.8 The greater role of home-based care in the United States must reflect the fact that Medicaid's stringent means-testing effectively shuts out middle-income families with elderly members who otherwise would be prepared to live in nursing homes. This does not occur in Canada, at least not to the same extent, and raises the basic issue of whether many of the elderly Americans who are taken care of at home might be better cared for in a nursing home, at least in one that maintained a highquality standard of care.

Some money is available for home care in Canada, but as Picard (2021, pp. 66-67) notes, coverage varies widely across provinces, with some provinces imposing monthly limits on the number of hours of care per care recipient and others a dollar limit. A family wishing to keep an elderly relative who needs substantial skilled care at home must pay for virtually all the care provided itself. These limitations on the available funding for home care must increase the demand for institutionalized care. Nonetheless, the share of Canada's elderly population who are cared for at home is substantially less than that in the United States. Some 900,000 are estimated to have received home care, compared with over 20 million in the United States (AARP 2020a). Again, if we scale up the Canadian figure by a factor of nine, the number of elderly persons cared for at home is less than one-half the U.S. number.

Germany

The provision of some form of long-term care in Germany, like the old-age pension, can be traced back to Bismarck's efforts to placate working-class unrest in the wake of the German Empire's rapid industrialization in the mid to late nineteenth century. Emanuel (2020) remarks on the relative continuity of German social policy over the subsequent cataclysmic upheavals of the First and Second World Wars and the separation and reunification of East and West Germany. East Germany's system was replaced upon reunification with the system of West Germany.

Contemporary Germany has a social insurance system but one with nearly universal coverage. The German states are not involved in financing the long-term care system. It is financed by a payroll tax that is currently set at 1.525 percent of wages up to a cap of €4.838 (about \$5,500) in 2021 matched by an equal contribution from the federal government, and it covers both workers and their dependents. 10 A period of two years of contributions is required before a benefit can be earned, but the system is effectively pay-as-you-go: the payroll tax rate has been increased several times in recent years to maintain a rough balance between current revenues and expenditures. Childless workers pay an extra 0.25 percent of their wages on the assumption that more of the care the average childless worker might need will be borne by the government, and less by his or her family. Able-bodied retirees may also contribute to the system, and the self-employed pay both the worker's and the government's share of the payroll tax, as is the case with Social Security in the United States.

Perhaps the key feature of the German system is that the federal government places fixed limits on the payments it will make to finance long-term care, with specific limits that increase with the degree or amount of care required. There currently are five different levels at which care is deemed to be provided, with a fixed Euro limit paid for each. As a result, any residual or remainder expense is borne by the recipients or their families. For some families, particularly those with a member needing institutional care, this arrangement can entail a significant burden, which is partially alleviated by social assistance from the local community. In 2019, about one-third of residents in long-term care facilities were receiving social assistance. The federal government recently set a limit on the expenditure that families have to bear for the care of their dependent older members. Local communities can require that the children of care recipients reimburse them at least partially for the assistance that local communities pay if the income of children exceeds a certain level, which has increased over the years and is now relatively high. This feature reduces the cost of the social assistance to the local community but must introduce a degree of complexity into the long-term care financing system.

The current system has an obvious benefit for the federal government in that it is not financially open ended. However, this same feature means that unless the payments the federal government makes are adjusted (increased) at reasonably frequent intervals, the burden borne by families will simply increase every year. Under the current policy, the federal government does in fact increase the benefits it pays at each of the five levels of care every three years. The system really requires that a delicate balance be struck between the welfare of families with members needing long-term care and the financial position of the federal government. In the German system, virtually all families with an elderly member being cared for at home are receiving either cash or in-kind benefits (like visits from nurses or other health care providers). Cash benefits may be used to compensate family members for the time they spend as caregivers or to pay non-family members. This is not the general rule in either Canada or the United States.

One feature of the German system that is not in either the Canadian or the U.S. system is the reliance on immigrants to provide care, typically unskilled care. This reliance has undoubtedly held down costs, but it has also raised concerns about the quality of care. That said, the long-term care industry is not the only sector of the German economy relying on labor from other European Union countries. For those families not depending entirely on outside help, whether from immigrant labor or not, the law requires employers to give caregivers in their employ up to 10 days of mostly paid leave for caregiving. A proposal to substantially increase the generosity of this benefit is under consideration.

LESSONS FOR THE UNITED STATES

The long-term care system in the United States has one clear advantage over that of most other countries, including Germany and Canada: it is less expensive. A possible additional advantage is that the federal government may have more control, albeit indirect, over its costs than other countries do over their costs. Although Medicaid is an entitlement system, one in which states are obliged to provide the statutory level of care for a resident who meets Medicaid's eligibility requirements, the states have some incentive to control costs and can do so by stinting on the quality of care and by limiting the optional services they provide. Apart from its lower costs, however, the U.S. long-term care system does not compare well with that of either Canada or Germany.

Long-term care coverage in both countries, especially Germany, is substantially broader than it is in the United States. In Germany, families providing care do not necessarily have all their costs covered, but home care is much better supported, and family members can be remunerated for their time and effort. In contrast, care by fam-

ily members in the United States is mostly unremunerated. German families are not subject to the stringent means-testing that Medicaid applies, so, while not free, home-based care in Germany is available to middle- and upper-income families.

As already noted, Canada's system has the advantage over the American system in that middle-income families can afford to pay for institutionalized care. That care is both price and quantity rationed, as the long waiting lists imply. The money available for care at home, although it is not means-tested like in the United States, is not plentiful, however. As Chapter 2 discusses, the public financing of home care can raise some thorny administrative issues. Germany appears to have managed this aspect of long-term care better than North America.

Notes

- 1. The qualification "on paper" is necessary because there is little or no point in establishing a system that builds up a surplus if that surplus simply finances other government expenditures.
- 2. See Chapter 6 in Emanuel (2020).
- 3. Neither California nor Illinois apply the special income rule, the second pathway for eligibility for Medicaid. Consequently, no one with an income of more than 138 percent of the federal poverty line qualifies.
- 4. Previously, some institutional long-term care had been financed by municipalities and charities. The province had also provided some financing, but the 1949 act substantially increased its financial and administrative role (Association of Municipalities of Ontario 2011).
- 5. Low-income earners can apply for subsidies, but these are available only for basic accommodation, with perhaps three or four beds to a room.
- 6. These rates are the U.S. dollar equivalent of the Canadian dollar rates reported in Picard (2021, p. 52), converted at the exchange rate of one U.S. dollar equals 1.33 Canadian dollars, prevailing on December 31, 2021.
- 7. The figures for the number of residents in congregate settings comes from Picard (2021).
- 8. In this respect, long-term care is like health care in Canada: it is partly rationed by quantity, and not by price. For example, most Americans with insurance in need of hip replacement do not typically have to wait many weeks to schedule an operation. In Canada, unless the case is deemed to be urgent, the wait can be much longer.

- 9. This section draws heavily on Veghte (2021). Bismarck's social programs were not particularly expensive at their inception. Relatively few Germans survived to the age of 65, when they became entitled to a pension and when disability would have begun to set in with a vengeance.
- 10. Euros have been converted to dollars at the exchange rate of one Euro equals \$1.1371, prevailing on December 31, 2021.
- 11. A qualification of this remark as it applies to the United States may be needed, in that undocumented immigrants are known to play a significant role in the unskilled workforce in nursing homes.

Chapter 4

The Impact of the COVID-19 Pandemic on Older Americans in Nursing Homes and Other Institutions

The devastating effect that COVID-19 had on the residents in America's nursing homes became apparent even in the early stage of the pandemic. In spring of 2020, news reports began drawing attention to the unusually high death rates occurring in nursing homes. As of April 25, 2021, a couple months after the start of large-scale distribution of vaccines in the United States, an estimated 132,000 nursing home residents had died of COVID-19, or roughly 1 in 10 of the total number of residents (CMS 2021). The exact number may never be known, in part because the reliability of the data differs from state to state. In some states, deaths of caregivers have been confounded with deaths of residents. In addition, the classification of cause of death by coroners' offices was not uniform across the country.

The pandemic has also caused deaths of residents in ALFs. The fatality rate in ALFs appears to be much less than the rate among nursing homes, but the data are less reliable, mainly because of differences among states in reporting requirements. It is uncertain what impact the pandemic has had on residents in CCRCs.

NURSING HOMES

The shocking death toll in U.S. nursing homes is the result of at least five different influences:¹

1) COVID-19 has disproportionately killed older people, especially the very old (75+). This tendency is evident in the

death rates by age of the population at large. As of May 6, 2021, the total number of deaths attributed to the pandemic was about 576,000 (CDC 2021). Of that number, about 80 percent are estimated to have been aged 65 or older, and most of these deaths were in people aged 75 or older.² The disproportionate impact of COVID-19 on the elderly is a global phenomenon. The elderly account for the lion's share of deaths in Canada, Germany, the United Kingdom, and other countries as well. That said, the death rate among Americans aged 65+ who are not living in nursing homes is estimated to be 0.6 percent, which makes it a fraction of the death rate in nursing homes.³

- 2) Unlike ALFs, nursing home residents typically have little privacy. Most share a room with another resident or are housed in dormitory-style quarters. Consequently, infections of all kinds have more opportunity to spread.
- 3) Caregivers would bring in the diseases from the broader community. As of April 25, 2021, infections among caregivers were estimated to be about 575,000, compared with an estimated total number of caregivers of about 945,000 in 2016 (NCHS 2019). Deaths among caregivers were estimated at about 1,900 (CMS 2021), which—assuming no growth in the number of caregivers in 2016–2019—implies a death rate of about 2 per 1,000. Making the not unreasonable assumption that virtually all caregivers were aged less than 65 years, their rate of death is about five times that of the population at large in this age range, and their rate of infection was about five times that of the overall population.⁴
- 4) Morbidity (the prevalence of disease and ill health in general) is higher in nursing homes than it is among the general population of the same age.

5) Sanitation standards in many nursing homes, as Chapter 1 discusses, have often been substandard, and the penalties imposed on nursing homes reporting deficiencies in sanitary practices have not been onerous.

The pandemic will almost certainly have a profound impact on the structure of the nursing home industry. The adjustment process has not yet played out, and we cannot draw a firm conclusion on the demand for beds. However, the number of residents is estimated to have dropped by 16 percent between January 2020 and January 2021, or by about 200,000. Demand for beds may never fully recover without the adoption of new policies assuring that adequate sanitation standards will be maintained. In the meantime, potential residents will be seriously considering the options of aging at home and, their financial situation permitting, choosing to move to an ALF. That said, the COVID-19-related deaths in these institutions will have had a depressing effect on demand for apartments at ALFs as well.

On the supply side of the nursing home market, institutions will probably choose to reconfigure their premises to offer their residents more privacy, and in any case may be obliged to do so by changes to the regulations that Medicaid eligibility payment requirements will impose on them. These structural changes will reduce the number of beds a home can offer and will increase their break-even cost per resident. The combined influences of contracting supply and demand (in economists' language, leftward-shifting supply and demand curves) will reduce the number of beds the industry can offer. The impact on the cost to residents is uncertain.

The dissemination of vaccines that began in January 2021 strongly favored the residents in nursing homes, as it should have. By early May 2021, about 1.4 million residents in nursing homes, ALFs, and similar institutions had been fully vaccinated. Death rates in these institutions have dropped by almost 90 percent in 39 reporting states (which include the country's most populous) in the first four months of 2021 (Chidambaram and Garfield 2021). Nonetheless, COVID-19-related deaths in nursing homes as of early May 2021 were accounting for about 5 percent of nationwide deaths, despite the small share of Americans who reside in these facilities.

The largely successful campaign to vaccinate nursing home residents appears to have effectively ended the devastation the pandemic has wrought among the country's nursing homes, although vaccine hesitancy remains a problem among caregivers, as reflected in the fact that less than half are fully vaccinated. But the COVID-19 pandemic is certainly not going to be the last pandemic that America and the world will face. What happened in the country's nursing homes must never be allowed to happen again. Before turning to the experience in ALFs and CCRCs, we briefly discuss aspects of the international experience of long-term care institutions with the pandemic.

EXPERIENCES IN OTHER COUNTRIES

At least three possible causes of the high death rates in America's nursing homes—age, limited privacy, and comorbidities—will be found in other countries as well. It may be useful to compare the U.S. experience with that of Canada, Germany, and the United Kingdom. In the United Kingdom, residents in care homes, which are similar to U.S. nursing homes, were estimated in 2016 to account for about 4 percent of the population aged 65 years and older, which is about twice as high as the share of nursing homes in the United States. Deaths of residents amounted to about 9 percent of the population of care homes, a figure that is lower than the U.S. share, although obviously still extremely high (see Table 4.1).

In Canada, the death rate of residents in long-term care homes and other institutions for the elderly are estimated to be about 3 percent of the population of these institutions.⁵ The share of deaths of residents in the Canadian equivalent of nursing homes in total COVID-19 deaths is estimated to be 69 percent of total deaths (Canadian Institute for Health Information 2021), which is much higher than the share

···· · · ···· · · · · · · · · · · · ·				
	United	United		
	States	Kingdom	Canada	Germany
Number of nursing home deaths	150,000	41,521	14,739	29,000
Deaths as a share of home residents (%)	11.5	8.8	3.4	3.6
Deaths as a share of 65+ population (%)	0.28	0.35	0.22	0.16
Deaths as a share of national deaths (%)	26.5	27.4	69.0	31.4
Number of nursing home residents	1,300,000	472,562	428,525	800,000
Population aged 65+ (in millions)	54.07	11.81	6.84	18.09
Total COVID-19 deaths	565,000	151,795	21,361	92,271

Table 4.1 Pandemic-Related Nursing Home Deaths in the United States, the United Kingdom, and Canada (March 2020-April 2021)

NOTE: In the United Kingdom, nursing homes are known as care homes; in Canada, as long-term care homes.

SOURCE: Canada: number of long-term care home residents is from 2016 census; figure for long-term care homes includes Canadian equivalent of ALFs; 65+ population is Statistics Canada estimate for July 1, 2020. U.S.: 65+: Census; total and nursing home deaths CDC; nursing home residents based on state figures. U.K.: National Health Service for number of residents; MHA.org.uk for percentage of pop. 65+ in homes. Germany: Population and total deaths: Statista; nursing home residents and deaths: newspaper reports.

in the United States or the United Kingdom, but this simply reflects the much higher death rates in the population at large of these two countries.6

The United Kingdom, like the United States, has vaccinated a relatively large share of its older population and especially care home residents. About half the country was partially vaccinated as of early May 2021, and one-quarter was fully vaccinated. In Canada, although elderly people have been favored in the distribution of vaccines, vaccination rates were initially much lower than in the United States, in part because Canada lacks a maker of vaccines. Rates subsequently recovered as the country obtained more vaccines from abroad, and because provincial vaccination campaigns have not encountered the same resistance as they have in the United States. The rate of death

in German nursing homes, based on the available data, appears to be similar to the rate in Canada.⁷

It is impossible to fully understand why the rates of death in nursing homes in the United States and the United Kingdom were so much higher than in Canada and Germany, but two possible influences might have been at work. First, the rate of infection in the general population was much higher in the United States than in other countries, and that was reflected in caregivers' infection rates and death rates. Second, Medicaid's means-tested character probably resulted in U.S. nursing home residents having lower incomes than those in Canada or Germany. To the extent that income correlates with health, it could be inferred that comorbidities played a greater role in the United States than it did in these two countries.

ASSISTED LIVING FACILITIES AND CONTINUING CARE RETIREMENT COMMUNITIES

As Chapter 1 explains, Medicaid requires states to supply substantial information to qualify for eligibility in the program, but the same does not apply to either ALFs or CCRCs. As a result, the information available on the impact of COVID-19 is extremely limited in the case of ALFs and basically nonexistent for CCRCs.

A study by the Kaiser Family Foundation in September 2020 reported data on cumulative infection and death rates in ALFs for 19 states through August 2020. Thirty-one states and D.C. were not reporting cases or deaths in ALFs. Of the 19 reporting states, 10 provided data for both June and August 2020. For the 14 states reporting through August 2020, cumulative deaths were about 2,600 (True et al. 2020). No information for the subsequent period on cases or deaths from these states is readily available.

The available data do not permit a calculation of the number of deaths per ALF resident, although the comparatively low number of deaths for the 14 reporting states, including California, Connecticut,

New York, Pennsylvania, and Texas—states that account for a large share of the population of ALF residents—suggest that it is substantially lower than the rate of death at nursing homes.

Subsequently, New York State began reporting deaths in ALFs, nursing homes, and other facilities at weekly intervals, with cumulative totals. In the first week of March 2021, deaths in ALFs accounted for 5 percent of cumulative total deaths in long-term care institutions of 15,430 (Long-Term Care Community Coalition 2021). Timely and detailed data for other states do not appear to be readily available. If, however, the situation in New York may be taken as representative of the whole country, the conclusion that the death rate among residents in ALFs is much lower than that at nursing homes is reinforced.

The limited data on CCRCs do not allow any firm conclusion to be drawn as to the impact of the pandemic on these facilities. It is likely, however, to have been similar in ALFs, and possibly less given their more spacious living arrangements. That said, CCRCs do include facilities for residents who can no longer care for themselves. To the extent that the living arrangements for these residents are like nursing homes, similar problems may arise.

In sum, we cannot offer a comprehensive explanation of the appallingly high death rates in U.S. nursing homes, but the higher rate of infection in the population at large, which affected the infection and death rates among caregivers, probably played a role. Higher comorbidity may also have contributed. The higher infection rate in the population at large was not within the control of nursing homes, but their response to it was, at least in part.

Notes

- 1. One very troubling aspect of the COVID-19 pandemic that affects younger survivors is so-called long COVID: the lingering and often debilitating side effects of the disease on survivors. See The Economist (2021).
- 2. The total number of deaths for which age at death is recorded is about 80 percent of the total number of deaths, apparently because the age

- at death of about one in five victims of COVID-19 could not be determined or was not recorded.
- 3. Americans aged 65 and older numbered about 54 million in 2019, according to the American Community Survey. With an estimated 1.3 million nursing home residents in 2019, the number living outside homes was about 53 million. If we assume that nursing home residents are all aged 65+ years, which is a reasonable approximation, COVID-19 related deaths were 0.6 percent of the population aged 65+ outside of nursing homes.
- 4. The death rate of the population less than 65 years old is estimated to be 0.04 percent.
- 5. The death rate in Canada is for all congregate facilities, not just for long-term care (nursing) homes. The rate for nursing homes would be higher than the rate shown in Table 4.1 if other congregate facilities were excluded. However, it would remain well below the U.S. rate.
- 6. The lower death rate at Canadian institutions does not in any way imply that the administration in these places has been satisfactory. Picard (2021) documents some shocking cases of mismanagement and neglect, including some egregious cases of what amounts to malpractice in one long-term care home in Montréal.
- 7. The figure for Germany shown in Table 4.1 is the sum of estimates for the German states taken from various sources. It should, therefore, be considered as representing only an order of magnitude.
- 8. The Canadian Institute for Health Information estimates that as of March 2021, about 25 caregivers had died in Canadian facilities. Even adjusted for the difference in population between Canada and the United States, this is a fraction of the number of deaths of caregivers in U.S. nursing homes.

Chapter 5

Recommendations for Policy

When considering policies for improving long-term care in the United States, any set of recommendations must be mindful of what is realistically attainable. Politics is the art of the possible, and as Voltaire observed, the perfect is the enemy of the good. That said, it might be useful to set out what a *good*, if not a perfect, system of long-term care might look like as a benchmark for reform. Setting a benchmark is not the same as establishing an edict that must be met to the letter. It is merely a way of giving reform a coherent and consistent set of objectives and pointing to areas where current practice is too far from the norm to really be acceptable.

With that in mind, this chapter proceeds with a summary account of what the author believes a good long-term care system would look like. It will then draw together the analyses of the chapters that have preceded it and briefly summarize the ways in which current practice meets or falls short of this benchmark. Finally, it will make recommendations to bring current practice closer to the norm described here, setting out three progressively more ambitious approaches to reform.

A THUMBNAIL SKETCH OF A GOOD SYSTEM

A good system should provide at least adequate long-term care to every citizen who needs it, in an appropriate setting. That setting may be an institution, the person's home, or a community care center. The cost of care at home may well be less expensive than institutional care, although this will depend on the level and nature of care that each care recipient needs. Persons suffering from severe dementia, for example, require around-the-clock care, which is much less expen-

sive if provided in a congregate setting than it would be at home. That said, some families may have the means to pay for intensive at-home care and may choose to do so.

Paying for care should not immiserate the care recipient or her family. The burden of good care on the public purse and therefore on the taxpayer should be minimized by ensuring that families that can afford it should pay for all or most of the care, or more generally, that the role of the public support should vary inversely with the income or wealth of the care recipient. Except for the very poor, the care recipient or his family should pay for at least part of his care.

A good system should minimize waste and fraud. To that end, public support of home care as well as institutional care should be adequately monitored, which will require that caregiving institutions provide appropriate and timely information on their activities and their budgets. In addition to adequate information, there must be incentives in place to effectively minimize poor-quality care or dangerous practices, especially in nursing homes. Monitoring care given in the home is likely more costly than monitoring institutional care, because of economies of scale.²

Frontline workers, both skilled and less skilled, should be adequately paid, trained, and vetted. The less-skilled workers in nursing homes, who may have the most contact with residents, need to be thoroughly trained in procedures that minimize the risk of exposing residents to infections, and understand the importance of helping more immobile residents to avoid bed sores.

A good system should also be as easy as is feasible given its inherent complexity for its users, actual and potential, to understand. Less misunderstanding of the rules that determine who is eligible for publicly supported care and the terms of those rules will make it easier for everyone to plan for this contingency. Information on publicly provided care should be published at regular and timely intervals.

Finally, any long-term care system involves politically sensitive trade-offs between the cost to the taxpayer, the quality of care, and the financial burden borne by low- and middle-income households.

Making that trade-off as easy as possible for the country's political system to deal with is another critical feature of a good system and will require that the incentives for efficiency and waste minimization be as well designed as possible. It will also require some general acceptance of the need for higher taxes, if higher taxes are needed, to allow the quality and coverage of long-term care to reach a norm.

HOW DOES THE CURRENT SYSTEM COMPARE TO THE BENCHMARK?

Chapter 3 places the U.S. system in the general category of a residual (i.e., means-tested) system. The U.S. system has the undoubted merit of being less expensive than the systems of any other advanced country. However, when compared to the benchmark we have just outlined, it is deficient in some crucial respects. Of particular concern are the following shortcomings:

• Coverage and means-testing. Unlike Germany and the Netherlands, where coverage of long-term care is either universal or nearly so, coverage in the United States is quite severely limited by means-testing. The result is that many middle-class families will not be able to avail themselves of it without first spending down their assets or engaging in estate planning (which is not cheap) to protect their assets from encroachment. Even then, they may fail their state's income test. Even in Canada, where coverage of institutional care or care at home is not universal, government subsidies bring institutional care into the reach of most middle-income families. The same is true of the United Kingdom, as the OECD study described in Chapter 3 demonstrates. More generally, the U.S. system can push families to the brink of poverty or require that they care for their infirm elderly at home when institutional care might be superior.

- Care at home. Caregivers at home receive little support from the government and as a result are often obliged to forgo remunerated work outside the home. The AARP's surveys report significant levels of burnout and psychological stress.
- homes currently provide comprehensive data, but there is an unnecessarily long lag in its dissemination. Data on ALFs is fairly comprehensive in some states, but skimpy or nonexistent in others. It is generally not made available promptly. Data on CCRCs is almost completely lacking. Regular reporting on care at home understandably does not take place, although AARP and others have produced very useful and quite comprehensive surveys of the number of caregivers and their recipients and other relevant information. This disparity in reporting across congregate institutions mainly reflects the fact that Medicaid is jointly financed by the federal government and the states, while ALFs and CCRCs are financed entirely by their residents.
- Understandability. No one who has studied or attempted to study Medicaid would seriously argue that its rules are easy to understand. The nature of long-term care would make the rules that applied to its public provision complex regardless of any efforts to simplify them. But the combination of the divided jurisdiction between the federal and state levels of government and the role of means-testing adds layers of complexity. It is no wonder, as Chapter 2 notes, that a large share of Americans over the age of 50 are confused about who pays for what.
- Incentives to rectify shortcomings in care in congregate facilities. This is almost certainly a more serious problem in nursing homes than it is in other congregate settings or care at home. Although nursing homes are required to report deficiencies in the care of their residents to their state by Medic-

aid, and appear to do so at the required frequency, sanctions for unhealthy and dangerous practices are rarely invoked. Although the data available on the impact of the pandemic in nursing homes or equivalent facilities in other countries is very limited, what there is makes clear that nursing homes in the United States were shockingly unprepared to deal with the risk of contagion and infection. The difference between the death rates in nursing homes in the United States and those of its northern neighbor is remarkable, despite the inadequacies that have been reported in Canada.

 Pay and work conditions of unskilled workers at nursing homes. Unskilled workers at nursing homes are typically paid somewhat less than \$15 per hour, the wage seen by many as a minimum living wage. It is questionable whether that rate is high enough to attract the dedicated workers that the job demands.

DIRECTIONS FOR REFORM

This section sets out three possible approaches to reform of the current system that would bring it into closer alignment with the model previously described. The first is the simplest and easiest to implement, at least from a technical point of view. It would maintain the structure of the status quo while addressing at least in part what the author believes are the worst failings of the current system. This first approach, assuming it had political support, would be the quickest reform to implement, but like the more ambitious versions that follow, it would require additional taxes to finance it. The second approach would add some features to the first and would significantly improve it. It would, however, require yet more taxpayer dollars. The third approach would require a root and branch reform of the current system. It would take some time to implement—along with its additional cost, it would require enormous political will, and a consensus that the current system is simply too broken to be fixed.

The First Approach

The first, simplest, and least costly approach would leave the current financial structure of Medicaid in place. The program would remain jointly financed by the federal government and the states, but it would be made more generous by making the means-testing rules less stringent, both as they apply to custodial care and as they apply to HCBS. This reform would simply bring public support to more middle-class families. How much more generous the program could be would obviously depend on the willingness of taxpayers to pay more for it. To reduce the effect of moral hazard, the program might be redesigned with a sliding scale that would eliminate the strict cut-offs of the current system. The addition of this sliding scale would be more complex than simply raising the asset and income tests.3 Another reform, albeit one that would require some change in the division of responsibility between the two levels of government, would be to make more uniform the rules that states apply to countable assets, although this might prove to be a bridge too far.

Simply making means-testing less stringent would entail its own complications. The reader will recall that there are three pathways to Medicaid eligibility: the original pathway, with an income ceiling depending on the state of 74 percent or 100 percent of the federal poverty line (which, in the 39 states including D.C. that have accepted the ACA, is now 138 percent); the special income rule, with a ceiling of three times the SSI level; and the medically needy rule, the most complex of the three. Raising the limit of the special income rule, which applies in 42 states and D.C. would be technically straightforward if politically contentious. The original pathway might be adjusted by a multiple of its current level in each state, like 1.25. Adjusting the limits of the medically necessary pathway would be the most complex because of the way the income floor varies from state to state. Similar considerations would apply to the asset ceiling that states impose.

One issue that might arise with this first approach, and which would undoubtedly be politically contentious, is the division of financing between the federal government and the states. Would the rules stay the same—a minimum of a 50 percent share and a maximum of a 77 percent share for the federal government, or would there be substantial pressure for the federal government to pick up more or even all the extra costs that the reform would entail? Even if the federal government agreed to increase its share of total costs, disagreement among the states as to what that increased share should be is quite possible. In sum, even an increase in public support with the basic structure of Medicaid financing left unchanged would entail some technical issues that would need to be worked out, and perhaps more importantly, would be difficult politically.

The Second Approach—The First Approach with an Add-on

The second approach would build on the first by including remuneration for the millions of unpaid relatives and non-family members who care for the elderly infirm in their homes. A system of payments to these caregivers is described in the last section of Chapter 2. Payment based on a strict system of periodic reports on the number of hours spent by caregivers was deemed to be infeasible. Instead, a presumptive system was proposed, where there might be annual inspections of a home where care is being provided and an assumed number of hours of caregiving assessed. This assessment could be based on an evaluation of the needs of the care recipient and the number of persons being cared for in each home. The annual assessment could be requested by the caregiver or his or her representative and need not generate much paperwork. In many cases, only one assessment would be needed, and assessments perhaps would not need to be carried out every year. Remuneration per hour could be some multiple of the state's minimum wage.

Even this simplified system for determining what a caregiver should be paid would pose its own complications. Should the assessment be done by a trained nurse, as is the case with assessments by insurance companies that issue long-term care policies when a policy-holder initiates a claim? That would require tens or hundreds of thousands of visits in each state, each costing perhaps three or more hours of a skilled person's time. Or could a system work if it was based on an honor system, where it would be enough to show that an elderly person was living at home and where caregivers made a declaration—which would have to be subject to severe criminal penalties if it proved to be false—that he or she was in fact providing the care they claimed?

Another issue would arise with respect to the income of the family undertaking the care and whether it should be considered in determining the remuneration of caregivers. One way to limit costs would be to place a limit on the income of a caregiving household related to the number of its members. Under a sliding scale approach, the rate of hourly remuneration would be phased out as income per household member increased. A related issue is whether the hourly income rate would be determined by the state or by the federal government. Clearly, the second "add-on" approach would require that additional technical issues be resolved. More importantly, and while there may be no point in stating the obvious, political resistance would be considerable because of its extra cost.

The Third Approach—Root and Branch Reform

The third approach would entail a complete overhaul of the current system. Arguably, it would also require a fundamental change in Americans' attitudes toward their government and its role in the economy. The reader does not need reminding that a policy like Medicare for all has already encountered strong political resistance, and so would the reform of long-term care described here. The third approach is offered not as a realistic reform option, but rather as an ideal that the country might someday wish to pursue. Furthermore, this account does not presume to be a blueprint, ambitious or otherwise, but a mere sketch of the direction of proposed reform. The

reader will recognize that the proposal borrows important elements from the German system described in Chapter 3.

The basic elements of the reform would be universal coverage with safeguards to ensure that households would contribute toward the cost of long-term care according to their means. Universal coverage would either be combined with or entail an end to state-bystate regulations and options. Like Medicare, all states would offer the same coverage. There would be no means-testing, and hence, no need for state-by-state ceilings on income or assets.5 This aspect of the reform would have the great virtue of making the system less difficult for the average American to understand and navigate.

The program could be financed from general tax revenues (the universal model that Chapter 3 describes) or from an addition to the payroll taxes that now finance Social Security and Medicare (the social insurance model). Moving to a payroll tax system would require some adjustment in the taxes levied by the federal government and the states, because it would relieve states of the need to contribute directly to Medicaid's costs.

Nursing homes and any other congregate institutions would continue to set prices as they currently do under present arrangements and would continue to be reimbursed by state governments. However, the financing source of these payments might change. Under the social insurance approach, the federal government would make payments to the states, which would have to either cede part of their tax base to the federal government or compensate it, at least in part for the extra expenditure it was undertaking. The same change of arrangements would need to apply for HCBS.

Residents in subsidized congregate facilities would be expected to pay the government a charge for their room, board, and maintenance that would be related to their household income. Residents from households below a certain level would pay nothing, and a sliding scale would be established for households at higher levels of income.

Care at home could be provided by paid caregivers or professional staff from outside the home, as it now is, or by family members. Government could continue to pay caregivers from outside the home under a HCBS program whose elements would apply to the whole country, albeit at rates of remuneration that would vary across states. The government would pay relatives and the other unpaid caregivers who now come to the home of infirm older persons, and as is the case with institutional care, would establish a sliding scale that would apply to the total budget each household could allocate to this type of care, leaving the choice of the source of care in the home up to the household. Caregivers from outside the home would be subject to the same rates of remuneration that now apply. Relatives and other unpaid caregivers would be subject to the presumptive scheme described in Chapter 2.

Medicaid in its present form is largely administered by the states. The federal government has a critical financing role, which as pointed out under this reform, would have to change. It also has an indirect supervisory role of nursing homes, given the reporting requirements it imposes on states for access to Medicaid financing and the penalties it can impose if the standards it sets for nursing homes and other congregate facilities it pays for are not met. A change in financial arrangements might mean that the federal government would lose the leverage it now has to make states impose adequate standards of care. This could be avoided if the states remained responsible for paying nursing homes and were subject to a federal withholding of the funds they would normally get if they failed in their disciplinary role.

The devil is always in the details, and there are a lot of details that this broad-brush presentation of a very fundamental reform has not provided. Moreover, even if the details could all be worked out perfectly, there is the question of what it would cost. As the section on unpaid caregivers discussed, paying these workers at rates that workers in their states receive for similar work would cost about 2 percent of GDP, amounting to about 10 percent of federal government revenues and 6 percent of general government revenues. It is well beyond the scope of this study to estimate the cost of relaxing means-testing: for example, the cost of increasing the ceiling under

the special income pathway from three to four or five times the SSI. But estimates like these would have to be made.

The epigraph from Pearl Buck at the beginning of the book is worth pondering when we reflect on the cost of improving the care that we give to our most vulnerable citizens. Perhaps it would be well worth the high cost. If this study inspires its readers to take Ms. Buck's words seriously, it will have accomplished its basic objective.

Notes

- 1. This is an ethical judgment, one that cannot be proved or disproved. A libertarian might posit that persons in need of care or their families should be responsible for the costs of that care, regardless of their income or wealth level. The definition of "adequate" is a related but separate issue. We might all agree on the need to support the elderly poor but differ over the quality of that support—the amount of living space they should have, the quality of the food they eat, and so on. In any case, if we believe in a societal obligation to support the elderly disabled and their families, we must reckon with the possible moral hazard that such support could create: the danger that households will not undertake any saving to meet the contingencies of old age.
- 2. There are economies of scale in the sense that with institutionalized care, those being cared for are under one roof, rather than being separated by possibly long distances.
- 3. With a sliding scale, the cut-off point of the income test would be raised with the applicant's income.
- 4. Yet another issue relates to the treatment of household members who are minors—would they count as an adult in numbering the members of a household? Economists have sometimes assumed that minors should be counted for less than one adult, on the grounds that their basic expenditure or consumption needs are less.
- 5. Some functions of government, notably primary and secondary education and policing, are most efficiently carried out at the regional (i.e., state) and local levels. Medicare, however, is undoubtedly administered more efficiently at the federal level, and there is no obvious reason why its rules should vary from state to state. And if that is true of health care, it should also be true of long-term care. The same is true of Social Security. The country does not have 51 Social Security Administrations. Parenthetically, one might also wonder why it needs 51 separate agencies (the 50 states and D.C.) to administer unemployment insurance.

84 Mackenzie

6. This assumes that care recipients are either in a facility or are being cared for at home. If in each budgetary period a care recipient were receiving care both at an institution and at home, a total budgetary envelop would apply.

References

- Administration for Community Living. n.d. *Older Americans Act.* Washington, DC: U.S. Department of Health and Human Services, Administration for Community Living.
- American Association for Long-Term Care Insurance. 2021. "Long-Term Care Insurance Facts—Data—Statistics—2021 Reports." Westlake Village, CA: American Association for Long-Term Care Insurance. https://www.aaltci.org/long-term-care-insurance/learning-center/ltcfacts-2021.php (accessed July 26, 2022).
- American Association of Retired Persons (AARP). 2019. "How Continuing Care Retirement Communities Work." Washington, DC: AARP. https://www.aarp.org/caregiving/basics/info-2017/continuing-care-retirement -communities.html (accessed July 25, 2022).
- 2020a. "Caregiving in the United States 2020." Washington, DC: AARP. https://www.aarp.org/ppi/info-2020/caregiving-in-the-united-states.html (accessed July 25, 2022).
- ——. 2020b. "Companion Report: Caregiving in the U.S.: A Focused Look at Family Caregivers of Adults Aged 50+." Washington, DC: AARP. https://doi.org/10.26419/ppi.00103.022 (accessed July 25, 2022).
- American Health Care Association/National Center for Assisted Living (AHCA/NCAL). 2021. "Facts and Figures." Washington, DC: AHCA/NCAL. https://www.ahcancal.org/Assisted-Living/Facts-and-Figures/Pages/default.aspx.
- A Place for Mom. 2021. "Your Complete Guide to Veterans' Benefits for Long-Term Care." https://www.aplaceformom.com/veterans-benefits -guide (accessed July 25, 2022).
- Association of Municipalities of Ontario. 2011. "Coming of Age: The Municipal Role in Caring for Ontario's Seniors. An AMO Paper on Long Term Care and Senior Services." Toronto: Association of Municipalities of Ontario.
- Belbase, Anek, Anqi Cen, and Alicia H. Munnell. 2021. "What Level of Long-Term Services and Supports Do Retirees Need?" Issue Brief 21. Boston: Retirement Research Center, Boston College.
- Breeding, Ray. 2018. "Understanding the Regulatory Process for CCRCs." Blog, April 3, https://mylifesite.net/blog/post/understanding-regulatory -process-ccrcs/ (accessed July 26, 2022).
- Brown, Jeffrey R., and Amy Finkelstein. 2009. "The Private Market for Long-Term Care Insurance in the United States: A Review of the Evidence." *Journal of Risk and Insurance* 76(1): 5–29.
- Bureau of Labor Statistics. 2021. "Occupational Employment and Wages, May 2021." Washington, DC: Bureau of Labor Statistics. https://www.bls.gov/oes/current/oes311131.htm (accessed September 28, 2022).

- Canadian Association for Long Term Care. 2020. Crisis Point: Addressing the Needs of Seniors Living in Long-Term Care—Federal Pre-Budget Submission. https://www.caltc.ca/wordpress/wp-content/uploads/2020/02/CALTC budget submission 2020.pdf (accessed July 26, 2022).
- Canadian Institute for Health Information. 2021. The Impact of COVID-19 on Long-Term Care in Canada—Focus on the First 6 Months. Ottawa: Canadian Institute for Health Information.
- Centers for Disease Control. 2021. "COVID Data Tracker—United States COVID-19 Cases and Deaths by State." Washington, DC: CMS. https://www.covid.cdc.gov (accessed July 25, 2022).
- Centers for Medicare and Medicaid Services (CMS). 2015. Nursing Home Data Compendium 2015 Edition. Washington, DC: CMS.
- ——. 2021. "COVID-19 Nursing Home Data." Washington, DC: CMS. https://www.Data.CMS.gov (accessed July 26, 2022).
- Chidambaram, Priya, and Rachel Garfield. 2021. "COVID-19 Long-Term Care Deaths and Cases Are at an All-Time Low, Though a Rise in Cases in a Few States May Be Cause for Concern." San Francisco: Kaiser Family Foundation.
- Cohen, Marc A. 2019. "Testimony on Enhancing the Affordability and Accessibility of Private Long-Term Care Insurance." Written Testimony before the Senate Finance Committee/Subcommittee on Health Care. November 20. https://www.finance.senate.gov (accessed September 28, 2022).
- Cohen, Marc A., Eileen J. Tell, Edward A. Miller, Ann Hwang, and Michael Miller. 2020. *Learning from New State Initiatives in Financing Long-Term Services and Supports*. Boston: Leading Age LTSS Center.
- Correia, Candice L., Richard Sayre, and Jessica C. Allen. 2017. "Avoiding the Medicaid Trap: A Step-by-Step Guide to Estate Planning." *Journal of Retirement* 5(2): 96–103.
- Cottle, Michelle. 2021a. "Nobody Wants to Live in a Nursing Home. Something's Got to Give." *New York Times*, August 1. https://www.nytimes.com/2021/08/01/opinion/aging-nursing-homes.html (accessed July 25, 2022).
- ——. 2021b. "Who Will Take Care of America's Caregivers?" *New York Times*, August 12. https://www.nytimes.com/2021/08/12/opinion/caretakers-elderly-home-health-aides.html (accessed July 25, 2022).
- Crook, Michael, and Ronald Sutedja. 2017. "Will Long-Term Care Ruin Retirement Plans?" *Journal of Retirement* 4(3): 42–50.
- Department of Veterans Affairs (VA). n.d. "Geriatrics and Extended Care: Paying for Long-Term Care." Washington, DC: VA. https://www.va.gov/geriatrics/pages/Paying_for_Long_Term_Care.asp (accessed July 25, 2022).
- ———. 2021. "VA pension rates for Veterans." Washington, DC: VA. https://www.va.gov/pension/veterans-pension-rates/ (accessed July 25, 2022).

- The Economist. 2021. "Long COVID—And Now for the Aftershock" (May 1). Emanuel, Ezekiel J. 2020. Which Country Has the World's Best Health Care? New York: Public Affairs.
- Favreault, Melissa, and Judith Dey. 2021. "Long-term Care Services and Supports for Older Americans: Risks and Financing, 2020." ASPE Research Brief. Washington, DC: Assistant Secretary for Planning and Evaluation.
- Genworth. 2020. Cost of Care Survey 2020. https://www.genworth.com/aging-and-you/finances/cost-of-care.html (accessed July 25, 2022).
- Giefer, Katherine G., and Tracy A. Loveless. 2021. *Benefits Received by Veterans and Their Survivors*. Current Population Reports. Washington, DC: U.S. Census Bureau.
- Gleckman, Howard. 2020. "Hundreds of Thousands of Nursing Home Residents Don't Need To Be There." *Forbes*, August 28. https://www.forbes.com/sites/howardgleckman/2020/08/28/hundreds-of-thousands-of-nursing-home-residents-dont-need-to-be-there/?sh=7846268e6c44 (accessed July 25, 2022).
- Goffman, Erving. 1961. Asylums: Essays on the Social Situation of Mental Patients and Other Inmates. New York: Anchor Books.
- Gottlich, Vicky. 2021. "Policy Note: Important Changes to Money Follows the Person." Washington, DC: Administration for Community Living.
- Government Accountability Office (GAO). 2010. "Older Americans: Continuing Care Retirement Communities Can Provide Benefits, But Not Without Some Risk." GAO 10-611. Washington, DC: GAO.
- ——. 2019. "Nursing Homes: Better Oversight Needed to Protect Residents from Abuse." GAO-20-259T. Washington, DC: GAO.
- ———. 2020. "Infection Control Deficiencies Were Widespread and Persistent in Nursing Homes Prior to the COVID-19 Pandemic." GAO-20576R. Washington, DC: GAO.
- Harrington, Charlene, Helen Carrillo, Rachel Garfield, MaryBeth Musumeci, and Ellen Squires. 2018. "Nursing Facilities, Staffing, Residents and Facility Deficiencies." San Francisco: Kaiser Family Foundation.
- Harrington, Charlene, Steve Preston, Leslie Grant, and James H. Swan. 1992. "Revised Trends in States' Nursing Home Capacity." *Health Affairs* 11(2). https://doi.org/10.1377/hlthaff.11.2.170 (accessed July 25, 2022).
- Hashiguchi, Tiago Cravo Oliveira, and Ana Llena Nozal. 2020. "The Effectiveness of Social Protection for Long-Term Care in Old Age: Is Social Protection Reducing the Risk of Poverty Associated with Care Needs?" OECD Health Working Paper No. 117. Paris: Organisation for Economic Co-operation and Development.
- Iezzoni, Lisa I., Naomi Gallopyn, and Kezia Scales. 2019. "Historical Mismatch between Home-Based Care Policies and Laws Governing Home Care Workers." *Health Affairs* 39(6). https://www.healthaffairs.org/doi/10.1377/hlthaff.2018.05494 (accessed July 25, 2022).

- Kaiser Family Foundation. 2015. "Long-Term Care in the United States: A Time-Line." San Francisco: Kaiser Family Foundation. https://www.kff .org/medicaid/timeline/long-term-care-in-the-united-states-a-timeline/ (accessed July 25, 2022).
- —. 2020. "State Health Facts—Total Number of Residents in Certified Nursing Facilities." San Francisco: Kaiser Family Foundation. http://kff .org/other/state-indicator/number-of-nursing-facility-residents/ (accessed July 25, 2022).
- Kaye, H. Stephen, Mitchell P. LaPlante, and Charlene Harrington. 2009. "Do Noninstitutional Long-Term Care Services Reduce Medicaid Spending?" Health Affairs 28(1). https://www.healthaffairs.org/doi/abs/10.1377/ hlthaff.28.1.262?.= (accessed July 25, 2022).
- Lepore, Jill. 2018. These Truths: A History of the United States. New York: W.W. Norton.
- Life Plans. 2017. Who Buys Long-Term Care Insurance? Twenty-Five Years of Study of Buyers and Non-Buyers in 2015-2016. Waltham, MA: Life
- Long-Term Care Community Coalition. 2021. New York COVID-19 Fatality Data: Nursing Homes & Adult Care Facilities. https://www .nursinghome411.org/ny-nursinghome-covid-data (accessed July 25, 2022).
- Mackenzie, George A. (Sandy). 2020. Is There a Retirement Crisis? An Exploration of the Issues. CFA Foundation Research Institute. https:// www.cfainstitute.org/en/research/foundation/2020/is-there-a-retirement -crisis (accessed July 25, 2022).
- Musumeci, MaryBeth, Meghana Amula, and Robin Rudowitz. 2021. "Voices of Paid and Family Caregivers for Medicaid Enrollees Receiving HCBS." Kaiser Family Foundation Issue Brief. San Francisco: Kaiser Family Foundation.
- Musumeci, MaryBeth, and Priya Chidambaram. 2020. "Key Questions about Nursing Home Regulation and Oversight in the Wake of COVID-19." Kaiser Family Foundation Issue Brief. San Francisco: Kaiser Family Foundation.
- Musumeci, MaryBeth, Priya Chidambaram, and Molly O'Malley Watts. 2019. "Medicaid Financial Eligibility for Senior and People with Disabilities: Findings from a 50-State Survey." Kaiser Family Foundation. Issue brief. San Francisco: Kaiser Family Foundation.
- National Center for Assisted Living. 2019. Assisted Living State Regulatory Review. Washington, DC: National Center for Assisted Living.
- National Center for Health Statistics (NCHS). 2019. "Long-term Care Providers and Services Users in the United States, 2015-2016. Analytical and Epidemiological Studies." Vital and Health Statistics 3(43). Washington, DC: U.S. Department of Health and Human Services.

- National Pace Association. n.d. "The History of PACE." Milford, DE: National Pace Association. www.npaonline.org (accessed July 25, 2022).
- Nordman, Eric. 2016. The State of Long-Term Care Insurance: The Market. Its Challenges and Future Innovations. Washington, DC: National Association of Insurance Commissioners and Center for Insurance Policy and Research.
- Organisation for Economic Co-operation and Development (OECD). 2019. Health at a Glance 2019: OECD Indicators. https://www.oecd-ilibrary .org/social-issues-migration-health/health-at-a-glance-2019 4dd50c09en (accessed July 25, 2022).
- PHI. 2020. Caring for the Future: The Power and Potential of America's Direct Care Workers. Bronx: PHL
- —. 2021. Direct Care Workers in the United States. Bronx: PHI.
- Picard, André. 2021. Neglected No More: The Urgent Need to Improve the Lives of Canada's Elders in the Wake of a Pandemic. Toronto: Penguin Random House.
- Reinhard, Susan C., Lynn Friss Feinberg, Ari Houser, Rita Choula, and Molly Evans. 2019. "Valuing the Invaluable: Charting a Path Forward." AARP Public Policy Institute: Insight on the Issues, November. https:// www.aarp.org/ppi/info-2015/valuing-the-invaluable-2015-update.html (accessed July 25, 2022).
- Rudowitz, Robin, Elizabeth Williams, Elizabeth Hinton, and Rachel Garfield. 2021. "Medicaid Financing: The Basics." Issue brief. San Francisco: Kaiser Family Foundation.
- Skračić, Izadora, Tyler Bond and Dan Doonan. 2020. Accessing Long-term Care Through Medicaid: The Safety Net for Seniors Facing Unmanageable Costs. Washington, DC: National Institute for Retirement Security.
- Smith, David Barton, and Zhanlian Feng. 2010. "The Accumulated Challenges of Long-Term Care." Health Affairs 29(1). https://www.health affairs.org/doi/abs/10.1377/hlthaff.2009.0507 (accessed July 25, 2022).
- True, Sara, Nancy Ochieng, Juliette Cubanski, Priya Chidambaram, and Tricia Neuman, 2020. "Overlooked and Undercounted: The Growing Impact of COVID-19 on Assisted Living Facilities." San Francisco: Kaiser Familv Foundation.
- Veghte, Benjamin W. 2021. Designing Universal Long-Term Services and Supports: Lessons from Germany and Other Countries. Washington, DC: National Academy of Social Insurance.
- Veghte, Benjamin W., and Alexandra L. Bradley. 2017. "Medicaid and Federal Funding Caps: Implications for Access to Health Care and Long-Term Services and Supports among Vulnerable Americans." Health Policy 13. Washington, DC: National Academy of Social Insurance.

- Warshawsky, Mark J., and Ross A. Marchand. 2017. "Improving the System of Financing Long-Term Services and Supports for Older Americans." *Journal of Retirement* 5(1): 48–68.
- Watts, Molly O'Malley, MaryBeth Musumeci, and Priya Chidambaram. 2020. "Medicaid Home and Community-Based Services Enrollment and Spending." Issue brief. San Francisco: Kaiser Family Foundation.
- Wilkinson, Richard, and Kate Pickett. 2011. *The Spirit Level—Why Greater Equality Makes Societies Stronger*. New York: Bloomsbury Press.
- Winklevoss, Howard E., and Alwyn V. Powell. 1984. Continuing Care Retirement Communities: An Empirical, Legal and Financial Analysis. Philadelphia: University of Pennsylvania, Wharton School, Pension Research Council.

Author

George A. (Sandy) Mackenzie holds a master's degree in philosophy in economics from Oxford University, as well as an undergraduate degree from Dalhousie University. He is the author of two books published by Cambridge University Press, Annuity Markets and Pension Reform (2006) and The Decline of the Traditional Pension (2010), and Is There a Retirement Crisis? An Exploration of the Current Debate (2020), published by the CFA Institute. Sandy worked for the International Monetary Fund for over 25 years, where he led technical assistance missions to countries in South and Central America, Central Europe, the Middle East, and francophone Africa on a variety of public financial issues, including public pension reform and social expenditure issues. Sandy also worked at the AARP, where he was their "pensions expert," and he was formerly the editor of the Journal of Retirement, which publishes papers by both academics and practitioners on the economic and financial aspects of retirement.

Index

Note: The italic letters n or t following a page number indicate a note or table, respectively, on that page. Double letters mean more than one such consecutive item on a single page.

AARP. See American Association of Retired Persons Activities of daily living (ADLs) help with, 14–15, 18, 23, 45 LTC insurance and, 43–44, 51n31 Medicaid coverage of, 7, 13, 25n7 required data from nursing homes on, 9, 25n11 Administration on Aging (AOA), social services grants administered by, 26n15 Adult day care services, HCBS and, 12–14 Affordable Care Act (2010), 9, 78 Community Living Assistance Services and Supports program in, 9–10, 25–26n12 expanded HCBS and spousal benefits, 12, 26n17 Medicaid expansion under, 12, 26n17,	means-tested approaches to health care with, 53–56, 58, 59, 75 Medicaid redesign and, 78–79, 80, 81, 83nn3–4 as Medicaid requirement, 1, 2, 21, 33–34, 35–36, 38, 49nn7–9, 50n15, 63 as VA pension requirement, 22–23 Assisted living facilities (ALFs), 76 accommodations in, 1, 14, 15, 16, 37, 66 ADL help for residents in, vs. nursing homes, 14–15 cost for, vs. nursing homes, 15, 67 COVID-19 and, 16, 67, 69t, 70–71, 72nn5–7 growth of, 2, 5 history of, 14–16 Medicaid role in, 1, 15–16, 37 Asylums (Goffman), institution-based
32, 33, 49 <i>n</i> 5 Aging in place movement, goal of,	care and, 25 <i>n</i> 6
27 <i>n</i> 23 Aid and Assistance Benefit, VA and, 22, 23	Block grants, federal-state interaction and, 32, 49 <i>n</i> 4 Buck, Pearl, quote, ix, 83
Alaska, caregiver wages in, 51n36 ALFs. See Assisted living facilities American Association for Long Term Care Insurance, as data source, 44 American Association of Retired Persons (AARP), statistics from, 6, 24nn2– 3, 46, 76 Americans with Disabilities Act (1990), HCBS and, 12 AOA. See Administration on Aging Arizona, asset test for Medicaid coverage, 49n10 Arkansas, asset test for Medicaid coverage, 34	California LTC in, 24, 49n11, 54–55, 63n3 regulatory framework for CCRCs in, 42, 51nn27–28 Canada care homes regardless of age and means-testing in, 58–59, 63nn5–7 COVID-19 and elderly in, 66, 68–70, 69t, 72n8, 72nn5–6, 77 GDP spending on LTC by, 56t, 59, 63 health care system in, 57–60, 62, 75 law and legislation, 57, 63n4 provincial responsibility for health-
Asset and income tests concerns about U.S., 44, 55, 76, 78	and LTC in, 57, 58, 63 <i>n</i> 4 provision of long-term care in, 2–3, 58, 59, 63, 63 <i>n</i> 8

Care homes, Canadian, 58–60, 63nn5–7 Caregivers COVID-19 infection rates of, 66, 70, 72n4, 72n8 immigrants as, 62, 64n11 paid, with low wages, 2, 42–43, 51n30 persons cared for by, 6, 24n2, 48, 84n6 public financing of, 47–48, 79–80, 81–82, 83n4 unpaid, and LTC financing, 45–48, 51nn34–36 unpaid, in home-based long-term care, 1, 5–6, 13, 21, 27n19, 46, 76 CCRCs. See Continuing care retirement communities Center for Insurance Policy and Research, as data source, 42, 51n29 Centers for Medicare & Medicaid Services (CMS) ADL nursing home data published by, 9, 25n11 infection control deficiency data from nursing homes, 10–11, 25n14 Nursing Home Data Compendium published by, 10, 25n13 Certified nursing assistants, 15, 18, 19tt, 20t Civil War, Confederate states' support for widows, 24n1 Clinton, Pres. William J., administration, reform initiatives by, 10, 25–26n12 CMS. See Centers for Medicare & Medicaid Services Community-based service (HCBS) for long-term care, 78 adult day care services as component of, 12–14 federal-state interaction and, 2, 5, 32, 49n4, 81–82 Medicaid and, 1, 21, 32–33, 45, 50n15, 56 Medicaid coverage eligibility for, 33, 49n6, 50n16	Dementia, 51 <i>n</i> 31 facilities for persons with, 15, 58,
Medicaid and, 1, 21, 32-33, 45,	Delaware, SSI percentage limit in, 49n11
	· · · · · · · · · · · · · · · · · · ·
49n6, 50n16	73–74
noninstitutional advent of, and home care, 11–14, 26 <i>nn</i> 15–18, 27 <i>n</i> 19	Denmark, GDP spending on LTC by, 56 Direct care workers
	demographics of, 19–20, 19t, 20t

Direct care workers, cont. For-profit sector, nursing homes in, 10, occupational groups of, 18-21, 19tt, 25n520t, 27n20 France, LTC in, 53, 54-55, 56t public assistance for, in poverty, Fundraising by CCRCs, 50n24 20-21, 20ttraining of, 21, 27n21, 74 Germany Disabled persons, 18, 37, 51*n*34 costs of LTC in, 54-55, 56t, 62, 63 Medicaid for, 31, 33, 48n1 COVID-19 in, 66, 68, 69t, 70, 72n7 support for, 12, 47 health care system in, 60-62, 64n11, VA benefits for, 22–23 national pensions under Bismarck in, District of Columbia (D.C.). See Washington, D.C. 60,64n9provision of long-term care in, as Economic cost, caregiving and, 47, model for U.S., 2, 3, 61-62, 81 51n36social insurance in, 53, 54, 60, 63*n*2, Education, 20 public financing of, 46, 51n35 Goffman, Irving, author on regimented Elderly persons, ix care, 25n6 abuse of, in nursing homes, 10, 11 Gross domestic product (GDP) affluent, 5, 14, 36, 55 LTC spending in, G-7 and Nordic caregivers of, 6, 24nn2-3 countries, 56, 56t disabled, 6, 7, 12, 37, 47 reform proposal for U.S. and, 82-83 indigent, 2, 26n15, 31, 48n1 infirm, 1, 6, 7, 15, 23, 24–25n4, 46, Hawaii, LTC in, 24, 33, 49n11 48, 65-66, 71-72n2 HCBS. See Community-based service, populations of, in congregate facilities for long-term care vs. homes, 1, 6t Health care systems by country, risks faced by, 10-11, 25n14, 37, 39, comparative analysis of, 53-64 44, 66, 77 Canada, 57–60, 63n2, 63nn4–8 Employment, community service and Germany, 60-62, 64nn9-11, 75 OAA, 26n15 lessons for the U.S., 62-63 England means-tested approaches to, 53-56 Elizabethan, and poor laws, 7, 25n4, residual and hybrid coverage, 53, 57 social insurance, 53-54, 63n1, 81 household costs of LTC in, 54-55 United States, 53, 54-55, 56, 56t, 57, Estate planning, LTC and, 38, 45, 50n20, 58-59, 60, 63n3, 63n8, 64n11 51n33 universal comprehensive coverage, 53, 54, 75, 81–83, 83*n*5 Federal government, 24n1 Health insurance, 31, 33, 36 federal-state interaction and, 2, 5, 7, Health Security Act (1993), as reform 11, 12, 26*n*17, 32, 49*n*4, 81–82 initiative, 10, 25–26*n*12 Medicaid LTC financing by, and Hill Burton Act (mid-1950s), nursing states, 1, 2, 8, 25n8, 31-33, 49n3, home construction and, 7 76, 79 Holland Finland, GDP spending on LTC by, 56 LTC spending by, 56, 57 Florida, regulatory framework for social insurance in, 53–54, 63*n*1, 75 CCRCs in, 42, 51n27 Home-based long-term care, 60, 76 cost of, vs. institutional care, 73-74

Home-based long-term care, <i>cont</i> . COVID-19 death rates in, <i>vs.</i> nursing homes, 66, 72 <i>n</i> 3 financing of, 1, 20, 21, 24 <i>n</i> 1, 46, 56, 81–82	private, for LTC, 2, 13, 24n1, 36–37, 42–45, 51nn29–33, 54 See also Health insurance Italy, GDP spending on LTC by, 56t
populations in, and HCBS vs. congregate facilities, 1, 5, 6t relative poverty and, 55, 74 tendency for, over institutional-based care, 54, 59	Japan, LTC in, 54–55, 56t Johnson, Pres. Lyndon B., administration, Medicaid coverage and, 8
See also Caregivers; Community- based service (HCBS) for long- term care	Kaiser Family Foundation, as modern LTC history resource, 5, 6 <i>t</i> , 10, 21, 24 <i>n</i> 1
Home health agencies, private funding	I I C C ITC
of, 13 Homes for the Aged Act (1949), Canada,	Labor force for LTC direct care workers, 18–21, 19tt, 20t,
57, 63 <i>n</i> 4	27 <i>n</i> 20
Hospitals, 57	paid, 18–21, 19tt, 20t, 27nn20–23
disproportionate share, (DSH) and	staffing, 15, 18–19
Medicaid, 32, 49n5	unpaid, 1, 5–6, 13, 27n19
Housebound Benefit, VA and, 22, 23	See also Caregivers; Certified nursing assistants; Nurses; Unskilled aides
Illinois, 49 <i>n</i> 11	Life care annuities, 45
household costs of LTC in, 54–55, 63n3	Life-Care LTC Protection Acts (1988), as reform initiative, 10, 25–26 <i>n</i> 12
medically needy rule in, 34–35, 49–50 <i>n</i> 13 regulatory framework for CCRCs in,	Long-term care (LTC) financing and provision of, 1–3, 5, 48, 62
42, 51 <i>n</i> 27	GDP spending on, by G-7 country,
Institution-based long-term care, 44	56, 56 <i>t</i>
basic information from, 2, 5, 10–11,	models for, 2–3, 61–62
74, 76 congregate facilities for, 1, 2, 6, 6 <i>t</i> ,	populations in congregate facilities vs. homes for, 1
25 <i>n</i> 6, 74, 83 <i>n</i> 2	private insurance for, and payouts, 2,
(see also Assisted living facilities	24 <i>n</i> 1, 42–43
(ALFs); Continuing care retirement communities (CCRCs);	probabilities of need for, 37, 50 <i>n</i> 17, 50 <i>n</i> 29
Nursing homes)	reform of, as progressive and
cost of, vs. home-based, 73–74, 83n2 eligibility for, Medicaid coverage, 33,	politically contentious, 3, 47, 79, 80
49n6	types of
financing of, 2, 24 <i>n</i> 1, 48, 63	(see Community-based long-term
Insurance, 45	care; Home-based long-term care;
data sources on, 42, 44, 51 <i>n</i> 29	Institution-based long-term care)
premiums, 31, 43–44	LTC, financing of, 31–51
premiums on sliding scale, 80, 81, 83 <i>nn</i> 3–4	CCRCs, 39–42, 50nn21–24, 51nn25– 28

Medicaid, 10 adult day care for elderly funded by, 12–14, 27 <i>n</i> 18 asset and income tests required by, 1, 2, 21, 35–36, 38, 50 <i>n</i> 15, 70 bias of, and moral hazard, 2, 8, 25 <i>n</i> 9, 78–79, 83 <i>n</i> 3 complexity of, 2, 7–8, 25 <i>nn</i> 7–8, 31, 35, 76
expansion of, 8, 12, 26 <i>n</i> 17, 32, 33,
49n5, 81–83, 84n6
financing long-term stays with, 1, 5,
7, 8, 36, 47 financing LTC eligibility for, 33–38,
37 <i>t</i> , 45, 49–50 <i>n</i> 13, 49 <i>nn</i> 6–12, 50 <i>nn</i> 14–20, 56, 67, 78
financing LTC with, 31–33, 44–45, 48nn1–2, 49nn3–5, 51n32, 54, 76
formal preadmission, screening
mandated by state governments, 8,
25 <i>n</i> 10
Washington (State) LTC system and, 5, 23–24
Medically needy rule, Medicaid
eligibility with, 34–35, 36, 37 <i>t</i> ,
49-50n13, 50n14, 50n16, 78
Medicare, 10, 80
age with premiums for, eligibility, 31,
36, 48 <i>n</i> 2
beliefs about, 44–45, 51 <i>n</i> 33
federal financing of short-term stays
with, 2, 7, 8, 14, 31, 36, 81, 83 <i>n</i> 5 Michigan, LTC policy in, 24
Miller trust, SSI percentage and, 34,
49 <i>n</i> 12
Minnesota, LTC policy in, 24
Missouri, SSI percentage limit in, 49 <i>n</i> 11
Money Follows the Person program, federal-state interaction and, 12,
26 <i>n</i> 17
Montana, SSI percentage limit in, 49 <i>n</i> 11
Moral hazard
bias of Medicaid and, 8, 25 <i>n</i> 9, 78–79, 83 <i>n</i> 3
LTC insurance and, 43–44
potential of, with home care and
HCBS, 11–12, 26 <i>n</i> 16
11CDS, 11-12, 20/110

National Association of Insurance	for-profit vs. nonprofit, 25n5, 35–36
Commissioners (NAIC), as data	history of, 2, 7–8, 24–25 <i>n</i> 4, 25 <i>nn</i> 5–9
source, 42, 44, 51 <i>n</i> 29	incentives to rectify shortcomings of,
National Building Museum, new use for,	2.9–10, 25–26 <i>n</i> 12, 74, 76–77
24 <i>n</i> 1	Medicaid and, 1, 2, 21, 35
National Center for Health Statistics	regulatory framework for, 5, 8–11,
(NCHS)	25–26 <i>n</i> 12, 25 <i>nn</i> 10–11, 26 <i>nn</i> 13–
ADL nursing home data published by,	14, 67
9,25n11	standard of care in, 8–9, 10–11,
data source for LTC residents, 6 <i>t</i> , 26 <i>n</i> 13	25n10, 26n13, 67
Nebraska, SSI percentage limit in, 49n11	Obama, Pres. Barack, administration,
Netherlands, The. See Holland	quality of nursing home care and, 9
New Hampshire, asset test for Medicaid	OBRA. See Omnibus Budget
coverage in, 34	Reconciliation Act, 1987
New York, 49 <i>n</i> 11	OECD. See Organisation for Economic
regulatory framework for CCRCs in,	Co-operation and Development
42, 51 <i>nn</i> 27–28	Ohio, regulatory framework for CCRCs
Nonprofit sector, nursing homes in, 10,	in, 42, 51 <i>n</i> 27
25 <i>n</i> 5	Old Age Assistance program, private
Nordic countries	nursing home industry and, 7,
health care in, with universal	25 <i>nn</i> 5–6
comprehensive coverage, 53, 54	Older Americans Act (OAA, 1965),
LTC spending in, compared to	social services provided by, 11,
industrial countries, 56, 57	26 <i>n</i> 15
North Carolina, LTC in, 42, 49n11	Omnibus Budget Reconciliation Act
North Dakota, SSI percentage limit in,	(OBRA, 1987), 8, 25 <i>n</i> 10
49n11	OOA. See Older Americans Act
Norway, GDP spending on LTC by, 56	Opportunity cost, caregiving with, 47
Nurses	Organisation for Economic Co-operation
assessment of caregivers by, 79-80	and Development (OECD)
registered and practical, in congregate	provision of long-term care in,
settings, 15, 18–19	countries, 2–3, 54–56, 56 <i>t</i> , 75
Nursing Home Data Compendium	
(CMS), $10, 25n13$	PACE See Programs for All-inclusive
Nursing Home Reform Act (1987), 8–9	Care for the Elderly
Nursing homes	Pennsylvania, regulatory framework for
accommodations in, 1, 37, 66, 67	CCRCs in, 42, 51 <i>n</i> 27
ADL help for residents in, vs. ALFs,	Pensions
14–15	German national, under Bismarck, 60,
annual reporting of basic data by,	64 <i>n</i> 9
10–11, 26 <i>nn</i> 13–14, 76	U.S. government, and qualifications
cost of, 38, 50 <i>n</i> 18, 74	for, 21, 48 <i>n</i> 2
COVID-19 and, in several countries,	Pepper recommendations (1990), as
3, 68–70, 69 <i>t</i> , 72 <i>nn</i> 5–8	reform initiative, 10, 25–26 <i>n</i> 12
COVID-19 and, in U.S., 6, 10, 16,	Personal care aides. see Unskilled aides

Poor Relief Act (1601), England, 7,

25n4, 57

65–68, 69*t*, 71–72*n*2, 71*n*1,

72nn3-4

Poorhouses, indigent elderly and, 2, 7, 24–25 <i>n</i> 4	before, and facilities for infirm elderly, 7, 24–25 <i>n</i> 4
Poverty	financing of, with payroll taxes, 47,
direct care workforce and, 20–21, 20 <i>t</i>	60, 81
relative, among household risks in	Social Security Disability Insurance, VA
OECD study, 54–55, 75	pensions and, 22
Private sector	Social welfare programs, 20–21, 20 <i>t</i> ,
	24 <i>n</i> 1
insurance for financing LTC in, 2, 13,	
24 <i>n</i> 1, 36–37, 42–45, 51 <i>nn</i> 29–33,	Special income rule
54	OECD findings of, in California and
nursing home industry, 7, 10, 25nn5–	Illinois, 55, 63 <i>n</i> 3
6, 67	SSI percentage limit and, 34, 36, 37 <i>t</i> ,
Programs for All-inclusive Care for the	49 <i>n</i> 11, 78, 82–83
Elderly (PACE) model, as state	State governments, 31, 62
option for avoiding institutions for	eschewing Medicaid LTC financing
home services, 13–14	by, 5, 23–24
Psychiatric hospitals, indigent elderly	federal-state interaction and, 2, 5, 7,
and, 24–25n4, 25n6	11, 12, 26 <i>n</i> 17, 32, 49 <i>n</i> 4, 81–82
Public assistance, definition, 20t	formal preadmission screening
Public sector	mandated by, 25n10
compensating caregivers by, 46-47,	home-based caregivers and, 1, 12-13
51 <i>nn</i> 34–35	26n16
institutional care used prior to home	Medicaid LTC financing by federal
care in, 10, 54	and, 1, 2, 8, 25 <i>n</i> 8, 31–33, 49 <i>n</i> 3, 76, 79
Quality of care, 44	psychiatric hospitals managed by,
CCRCs and, 17, 18	24–25 <i>n</i> 4
as required annual report by nursing	regulation of LTC by, 5, 7, 10, 43
homes, 10–11, 26 <i>n</i> 13	Supplemental Security Income (SSI), as
standard of care and, 8–9, 25 <i>n</i> 10, 48,	support qualifier, 22, 33, 34,
59, 62	49 <i>nn</i> 11–12, 78
,	Sweden, GDP spending on LTC by, 56
trade-offs between, and costs, 74–75	Sweden, GDP spending on LTC by, 56
Risks	Taxes
infection, 10-11, 25n14, 66, 77	general revenue, for LTC, 75, 77, 81
insurance amelioration of, 38, 43-44,	payroll, for LTC, 47–48, 60, 64 <i>n</i> 10,
45	81
longevity, 39–40, 44, 50 <i>nn</i> 21–22,	Tennessee, medically needy rule in,
51n25	34–35, 49–50 <i>n</i> 13
relative poverty among, 54–55, 76	Texas
relative poverty among, 34 33, 70	medically needy rule in, 34–35,
Scandinavia. See Nordic countries	49–50 <i>n</i> 13
Shakespeare, William, quote, ix	regulatory framework for CCRCs in,
	42, 51 <i>nn</i> 27–28
Social Security Act (SSA, 1935) age of eligibility for, benefits, 31,	
age of eligibility for, benefits, 31, $48n2$	Title XX (1975), SSA amendment and
	social services, 11
amendments to, for social services by, 7, 11, 26 <i>n</i> 15	Trade-offs, benchmark, 74–75

United Kingdom (UK)	Veterans
COVID-19 in, 66, 68–70, 69t	Canadian, hospitals, 57
GDP spending on LTC by, 56t	non-pension benefits for, 22, 23,
provision of long-term care in, 3, 75	27 <i>n</i> 24
residual and means tested health care	retirement pensions for, 22, 24n1
system in, 53, 57	Voltaire, François-Marie, quote, 73
See also England	, , , , , , , , , , , , , , , , , , , ,
United States (U.S.)	Wages
provision of LTC in, 1–2, 3, 55, 56t,	paid caregivers and, 2, 13, 51n30,
62–63	51n36, 79–80
social insurance and residual health	unskilled aides, 15, 19-20, 21, 27n22,
care system in, 53–54, 63 <i>n</i> 1	74, 77
weaknesses of LTC in, 1-2, 75-77	Washington, D.C., pension
See also states within, e.g., California;	administration from, 24n1
Illinois; Texas	Washington (State), LTC system of
U.S. Congress, 22	Medicaid and, 5, 23-24
failure to pass reform initiatives, 10,	vesting and residency requirements,
25–26 <i>n</i> 12	23, 27 <i>n</i> 25, 54
U.S. Dept. of Health, Education and	West Virginia, medically needy rule in,
Welfare, 8	34–35, 49–50 <i>n</i> 13, 50 <i>n</i> 14
U.S. Dept. of Veterans Affairs (VA), LTC	Wisconsin, regulatory framework for
program of, 22–23, 27 <i>n</i> 24	CCRCs in, 42, 51 <i>n</i> 27
U.S. law and legislation	
disabled persons, 12	Yeats, W.B., quote, ix
long-term care initiatives, 7, 9–10,	
24 <i>n</i> 1, 25–26 <i>n</i> 12	
nursing homes, 7, 8–9, 25 <i>n</i> 10	
social services, 11, 26 <i>n</i> 15	
Universal comprehensive coverage	
health care in Nordic countries with,	
53, 54	
Medicaid reform with, 81–83, 83 <i>n</i> 5	
near, in Germany, 60, 75	
Unskilled aides	
home health and personal care aides	
as, 18, 19tt, 20t, 21, 27n20, 37,	
50n17	
immigrants as, 62, 64 <i>n</i> 11	
training of, 21, 27 <i>n</i> 21, 74	
wages of, 15, 74, 77	
VA (U.S. Dept. of Veterans Affairs),	
22–23, 27 <i>n</i> 24	
Vaccines, COVID-19, and nursing	
homes, 3, 6, 10, 67–68	
Vermont	
medically needy rule, 34-35, 49-	
50 <i>n</i> 13	

About the Institute

The W.E. Upjohn Institute for Employment Research is a nonprofit research organization devoted to finding and promoting solutions to employment-related problems at the national, state, and local levels. It is an activity of the W.E. Upjohn Unemployment Trustee Corporation, which was established in 1932 to administer a fund set aside by Dr. W.E. Upjohn, founder of The Upjohn Company, to seek ways to counteract the loss of employment income during economic downturns.

The Institute is funded largely by income from the W.E. Upjohn Unemployment Trust, supplemented by outside grants, contracts, and sales of publications. Activities of the Institute comprise the following elements: 1) a research program conducted by a resident staff of professional social scientists; 2) the Early Career Research Award program, which provides funding for emerging scholars to complete policy-relevant research on labor-market issues; 3) a publications program and online research repository, which provide vehicles for disseminating the research of staff and outside scholars; 4) a regional team that conducts analyses for local economic and workforce development; and 5) the Employment Management Services Division, which administers publicly funded employment and training services as Michigan Works! Southwest in the Institute's local four-county area.

The broad objectives of the Institute's activities are to 1) promote scholarship and evidence-based practices on issues of employment and unemployment policy, and 2) make knowledge and scholarship relevant and useful to policymakers in their pursuit of solutions related to employment and unemployment.

Current areas of concentration for these programs include the causes, consequences, and measures to alleviate unemployment; social insurance and income maintenance programs; compensation and benefits; workforce skills; nonstandard work arrangements; and place-based policy initiatives for strengthening regional economic development and local labor markets.



Cover Design: carol a.s. derks | derkstudio

